PAGE 1

FORM 10-0

SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 29549

(X) QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended September 30, 1997

or

() TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the transition period from _____ to____

Commission File Number 1-3229

NORTHROP GRUMMAN CORPORATION (Exact name of registrant as specified in its charter)

DELAWARE
(State or other jurisdiction of incorporation or organization)

No. 95-1055798 (I.R.S. Employer Identification No.)

1840 Century Park East, Los Angeles, California 90067 (address of principal executive offices)

(310) 553-6262

(Registrant's telephone number, including area code)

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days.

Yes x No

APPLICABLE ONLY TO CORPORATE ISSUERS

Indicate the number of shares outstanding of each of the issuer's classes of common stock, as of the latest practicable date.

Common Stock outstanding as of October 27, 1997

66,785,749 shares

Northrop Grumman Corporation and Subsidiaries

Part I. Financial Information Item 1. Financial Statements

CONSOLIDATED CONDENSED STATEMENTS OF INCOME

	Three months ended September 30,		Nine months e September	
Dollars in millions, except per share	1997	1996	1997	1996
Net sales Cost of sales	\$2 , 297	\$2,172	\$6,643	\$6,194
Operating costs	1,870	1,770	5,329	4,904
Administrative and general expenses	222	226	680	746
Operating margin	205	176	634	544
Other, net	25	20	30	44
Interest expense	(68)	(69)	(197)	(197)
Income before income taxes	162	127	467	391
Federal and foreign income taxes	64	49	177	151
Net income	\$ 98	\$ 78	\$ 290	\$ 240

Weighted average shares outstanding, in millions	66.7	66.3	66.7	61.3
Earnings per share	\$ 1.46	\$ 1.18	\$ 4.35	\$ 3.90
Dividends per share	\$.40	\$.40	\$ 1.20	\$ 1.20

CONSOLIDATED CONDENSED STATEMENTS OF FINANCIAL POSITION

Dollars in millions	September 30, 1997	December 31, 1996
Assets		
Cash and cash equivalents	\$ 48	\$ 123
Accounts receivable, net of progress payments of \$3,300 in 1997 and \$2,721 in 1996	1,574	1,453
Inventoried costs, net of progress payments of \$497 in 1997 and \$533 in 1996	1,315	1,053
Deferred income taxes	78	78
Prepaid expenses	78	68
Total current assets	3,093	2,775
Property, plant and equipment	3,230	3,195
Accumulated depreciation	(1,850)	(1,783)
	1,380	1,412
Goodwill, net of accumulated amortization of	0.444	0 450
\$220 in 1997 and \$150 in 1996	3,444	3,470
Other purchased intangibles, net of accumulated amortization of \$185 in 1997 and \$116 in 1996	919	988
Deferred income taxes	404	520
Prepaid pension cost, intangible pension asset	101	020
and benefit trust funds	392	229
Investments in and advances to		
affiliates and sundry assets	146	251
	5,305	5,458
	\$ 9 , 778	\$ 9,645

Dollars in millions	September 30, 1997	December 31, 1996
Liabilities and Shareholders' Equity		
Notes payable to banks	\$ 168	\$ 228
Current portion of long-term debt	200	200
Trade accounts payable	440	477
Accrued employees' compensation	422	357
Advances on contracts	289	230
Income taxes payable, including deferred		
income taxes of \$568 in 1997 and \$621 in 1996	622	646
Other current liabilities	529	531
Total current liabilities	2,670	2,669
Long-term debt	2,820	2,950
Accrued retiree benefits	1,671	1,624
Other long-term liabilities	54	59
Deferred income taxes	60	61
Paid-in capital		
Preferred stock, 10,000,000 shares		
authorized; none issued		
Common stock, 200,000,000 shares		
authorized; issued and outstanding:		
1997 66,755,872; 1996 66,527,262	789	784
Retained earnings	1,718	1,502
Unfunded pension losses, net of taxes	(4)	(4)
	2,503	2,282
	\$ 9,778	\$ 9,645

CONSOLIDATED CONDENSED STATEMENTS OF CASH FLOWS

Dollars in millions	Nine months ended	September 30,
DOTTALS IN MITTIONS	1997	1990
Operating Activities		
Sources of Cash		
Cash received from customers		
Progress payments	\$1,646	\$1,652
Other collections	4,909	4,527
Income tax refunds received	1	12
Interest received	4	10
Other cash receipts	4	5
Cash provided by operating activities	6,564	6,206
Uses of Cash		
Cash paid to suppliers and employees	6,039	5,436
Interest paid	189	123
Income taxes paid	47	68
Other cash disbursements	19	
Cash used in operating activities	6,294	5 , 627
Net cash provided by operating activities	270	579
Investing Activities		
Payment for purchase of Westinghouse ESG, net of cash acquired	d	(2,885)
Additions to property, plant and equipment	(141)	(131)
Proceeds from sale of property, plant and equipment	48	24
Proceeds from sale of affiliates/operations	19	18
Funding of retiree benefit trust		(25)
Proceeds from sale of marketable securities		9
Other investing activities	(8)	
Net cash used in investing activities	(82)	(2,973)
Financing Activities		
Borrowings under lines of credit	421	2,613
Repayment of borrowings under lines of credit	(462)	(616)
Proceeds from issuance of long-term debt		1,000
Principal payments of long-term debt	(150)	(832)
Proceeds from issuance of stock	8	
Dividends paid	(74)	
Other financing activities	(6)	
Net cash provided by (used in) financing activities	(263)	•
Increase(decrease) in cash and cash equivalents	(75)	
Cash and cash equivalents balance at beginning of period	123	
Cash and cash equivalents balance at end of period	\$ 48	\$ 158

	Nine months		Septembe	•
Dollars in millions		1997		1996
Reconciliation of Net Income to Net Cash				
Provided by Operating Activities				
Net income	\$	290	\$	240
Adjustments to reconcile net income to net cash provided				
Depreciation		164		164
Amortization of intangible assets		139		117
Common stock issued to employees		1		
Loss on disposals of property, plant and equipment		3		14
Retiree benefits cost(income)		(35)		23
Decrease(increase) in				
Accounts receivable		(524)		13
Inventoried costs		(189)		(53)
Prepaid expenses		(10)		(26)
Increase(decrease) in				
Progress payments		368		(30)
Accounts payable and accruals		40		144
Provisions for contract losses		14		(26)
Income taxes		134		122
Retiree benefits		(134)		(130)
Other transactions		9		7
Net cash provided by operating activities	\$	270	\$	579
Noncash Investing Activities				
Purchase of businesses				
Fair value of assets acquired			\$	3,948
Cash paid				(2,890)
Liabilities assumed			\$	1,058

CONSOLIDATED CONDENSED STATEMENTS OF CHANGES IN SHAREHOLDERS' EQUITY

Dollars in millions	Nine months	ended 1997	September30, 1996
Paid-in Capital At beginning of year Stock issuance	\$	784	\$ 273 493
Treasury stock transactions Employee stock awards and options exercised,		(5)	(6)
net of forfeitures		10 789	9 769
Retained Earnings			
At beginning of year		1,502	1,326
Net income		290	240
Cash dividends		(74)	(65)
	:	1,718	1,501
Unfunded Pension Losses, Net of Taxes		(4)	(12)
Total shareholders' equity	\$:	2,503	\$2,258

SELECTED INDUSTRY SEGMENT INFORMATION

Dollars in millions	Three mont Sept 1997	ths ended tember30, 1996	Nine mont. Sept. 1997	hs ended ember30, 1996
Net Sales				
Aircraft	\$1 , 186	\$1 , 015	\$3 , 243	\$2 , 996
Electronics	966	1,010	2,927	2,748
Information Technology and Services	238	222	744	668
Intersegment sales	(93)	(75)	(271)	(218)
	\$2 , 297	\$2 , 172	\$6 , 643	\$6 , 194
Operating Profit				
Aircraft	\$ 183	\$ 156	\$ 460	\$ 395
Electronics	46	49	223	237
Information Technology and Services	22	26	69	50
Total operating profit	251	231	752	682
Adjustments to reconcile operating				
profit to operating margin:				
Other income included above	(4)	(7)	(7)	(9)
State and local income taxes	(2)	(6)	(7)	(28)
General corporate expenses	(32)	(31)	(92)	(90)
Mark-to-market restricted stock rights	(8)	(11)	(12)	(11)
Operating margin	\$ 205	\$ 176	\$ 634	\$ 544
Contract Acquisitions				
Aircraft	\$ 751	\$ 701	\$2 , 659	\$2,518
Electronics	467	546	2,811	5,045
Information Technology and Services	156	196	585	653
Intersegment acquisitions	(83)	(70)	(272)	(232)
	\$1 , 291	\$1 , 373	\$5 , 783	\$7 , 984
Funded Order Backlog				
Aircraft			\$ 6,459	\$ 6,535
Electronics			5,043	5,060
Information Technology and Services			353	423
Intersegment backlog			(48)	(49)
			\$11 , 807	\$11,969

Northrop Grumman Corporation and Subsidiaries

NOTES TO CONSOLIDATED CONDENSED FINANCIAL STATEMENTS

Basis of Presentation

The accompanying unaudited consolidated condensed financial statements have been prepared by management in accordance with the instructions to Form 10-Q of the Securities and Exchange Commission. They do not include all information and notes necessary for a complete presentation of financial position, results of operations, changes in shareholders' equity, and cash flows in conformity with generally accepted accounting principles. They do, however, in the opinion of management, include all adjustments (all of which were normal recurring accruals) necessary for a fair statement of the results for the periods presented. The financial statements should be read in conjunction with the Notes and Independent Auditors' Report contained in the company's 1996 Annual Report.

Business Combinations

Effective August 1, 1997, the company consummated the merger of its wholly owned acquisition subsidiary with and into Logicon, Inc., a leading defense information technology and services company. Each share of Logicon's common stock was converted to .6161 of a share of the company's common stock. Approximately 8.6 million shares of the company's common stock were issued for Logicon's common stock. The merger is accounted for as a pooling of interests and, accordingly, the accompanying financial statements have been retroactively restated for all periods presented to include the results of Logicon.

On July 3, 1997, the company announced that it had entered into a definitive agreement with Lockheed Martin Corporation to combine the companies. Under terms of the agreement, Northrop Grumman shareholders will receive 1.1923 shares of Lockheed Martin common stock for each share of Northrop Grumman stock. The transaction which is subject to shareholder approval and U.S. Government regulatory review, is estimated to be completed in the first quarter of 1998.

Inventoried Costs

The company's inventoried costs consist primarily of work in process related to long-term contracts with customers.

Earnings Per Share

In February 1997, Financial Accounting Standards Board Statement 128 - Earnings per Share was issued. This new standard becomes effective for financial statements for periods ending after December 15, 1997. Under the new standard, earnings per share would have been reported as follows:

	Three months	ended Se	eptember 30,	Nine mo	nths	ended S	September	30,
		1997	1996			19	997	1996
Earnings per share		\$1.46	1			\$4.		3.90
Diluted earnings pe	er share	\$1.44	\$1.14			\$4.	.27 \$	3.81

Item 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF THE COMPANY'S FINANCIAL CONDITION AND THE RESULTS OF ITS OPERATIONS

Effective August 1, 1997, the company consummated the merger with Logicon, Inc., a leading defense information technology and services company. Because the merger is being accounted for as a pooling of interests, financial data for all periods have been retroactively restated to include the results of Logicon. Logicon financial data are included in the information technology and services industry segment along with similar Northrop Grumman business previously classified in the aircraft and electronics business segments.

Sales were 16 percent higher in the third quarter of 1997 than the third quarter of 1996. Sales rose 17 percent in the nine months of 1997 versus the comparable period of 1996. Comparative results for 1996 include only seven months of operations of the defense electronics and systems business of Westinghouse, which was acquired March 1, 1996 and is now operated as the Electronics Sensors and Systems Division (ESSD).

Aircraft segment sales increased in the third quarter and first nine months of 1997 versus comparable periods of 1996 as a result of greater deliveries of shipsets for Boeing jetliners and the Gulfstream V business jet. The third quarter and nine month periods also benefited from increased B-2 spares and support sales. These increases were partially offset by decreased F/A-18 and B-2 production sales.

The electronics segment sales increase for the first nine months of this year over the same period last year is due to the inclusion of ESSD operations for the full nine months of 1997 versus seven months in 1996. Sales by major program/business area and units delivered were:

	Thre	e months	Nin	e months
<pre>\$ in millions</pre>	1997	1996	1997	1996
B-2	\$ 430	\$ 383	\$1,190	\$1,271
Surveillance Aircraft (E-8 Joint Stars, E-2)	222	250	745	811
F/A-18	144	185	396	521
Boeing Jetliners	211	160	636	421
Airborne Radar	195	172	491	388
ECM	95	87	288	286
Marine	154	140	426	311
C-17	73	73	215	194
Space	63	97	227	214
Airspace Management	65	58	194	171
Information Technology and Services	234	224	733	669
All Other	411	343	1,102	937
	\$2 , 297	\$2 , 172	\$6,643	\$6,194
	Thre	e months	Nin	e months
Units	1997	1996	1997	1996
747	10	9	32	21
F/A-18 C/D	8	17	26	49
F/A-18 E/F	0	0	0	3
C-17	2	2	6	5
B-2	2	1	2	4

Pension income, which is included in operating profit, was \$32 million for the third quarter and \$99 million for the first nine months of 1997 as compared with \$15 million and \$32 million for the same periods of 1996. Substantially all of the pension income is assigned to the aircraft segment.

The amount and rate of operating profit earned on sales increased in the aircraft segment in the third quarter and first nine months of 1997 as compared with the same periods of 1996. Aircraft segment operating profit benefited in the third quarter and first nine months of 1997 from increased Boeing jetliner and B-2 spares and support sales and an increase in pension income. These improvements were partially offset by lower F/A-18 operating margin on reduced sales. The third quarter of 1997 benefited from the delivery of two B-2 equivalent units compared to one in the third quarter of 1996. The nine month period of 1997 included two B-2 unit deliveries as compared to four deliveries in the same period of 1996. Last year's third quarter results benefited from the settlement of a claim involving productivity improvements on the F/A-18. Last year's nine month results included a \$25 million charge related to the company's work for Fokker Aircraft N.V., which declared bankruptcy in March 1996.

The amount and rate of operating profit earned on sales decreased in the electronics segment in the third quarter and first nine months of 1997 as compared with the same periods last year. Third quarter 1997 operating profit was reduced \$53 million by a cumulative margin rate adjustment on the E-8 Joint STARS. The charge was related principally to an increase in the number of hours required to remanufacture Boeing 707 aircraft. This reduction was offset partially by settlement of a claim involving work performed in the 1980's on the MX missile Interface Test Adapter (ITA), which resulted in an \$8 million increase in operating margin and \$12 million in interest income. Last year's third quarter included a \$29 million writedown of a claim related to avionics work performed by the former Grumman Corporation prior to its acquisition by Northrop in 1994, and a \$9 million cumulative downward margin rate adjustment to inspect and replace incorrect rivets installed on the E-8 Joint STARS aircraft. For the first nine months of 1997, electronics segment operating profit benefited from the inclusion of ESSD operations for the full period of 1997 versus seven months in 1996. Electronics operating profit was reduced by a \$13 million pretax charge recorded in the first quarter of 1997 related to an increase in the cost estimate to complete the company's work on the Directed Infrared Countermeasures (DIRCM) program.

Operating profit in the information technology and services business segment in the third quarter was down slightly from the same period a year ago. Last year's third quarter benefited from a claim settlement related to the TA-4J aircraft. Information Technology and Services business segment sales in the current quarter increased as a result of greater business volume within the company's Logicon subsidiary.

Other income for the third quarter of 1997 benefited by \$24 million from the recovery of a portion of a charge recorded in 1996 related to the sale of the Perry, Georgia, facility and by \$12 million in interest income related to the MX missile ITA claim settlement. Costs associated with the Logicon merger reduced other income and net income by \$18 million. Other income for last year's third quarter and nine month periods included gains of \$6 million and \$18 million, respectively, from sales of an equity investment in a manufacturer of high technology equipment.

Interest expense for the third quarter and first nine months of 1997 was level with the same periods of last year.

In June 1996 the company issued approximately 8 million shares of common stock in a public offering. The \$493 million in net proceeds from the issuance were used to pay down long-term debt. During the first nine months of 1997, \$270 million of cash was generated from operations versus \$579 million in the first nine months of 1996. The decrease is primarily due to the increase in working capital for the Boeing jetliners, Gulfstream V business jet and JSTARS programs in support of increased production levels. Cash generated from operating activities for the year is expected to be more than sufficient to finance capital expenditures and dividends. The company's liquidity and financial flexibility will continue to be provided by cash flow generated by operating activities, supplemented by the unused borrowing capacity available under the company's credit agreement and other short-term credit facilities.

In July 1997 the company entered into a definitive agreement with Lockheed Martin Corporation to combine the two companies. Under terms of the agreement, Northrop Grumman shareholders will receive 1.1923 shares of Lockheed Martin common stock for each share of Northrop Grumman stock. The transaction, which is subject to shareholder approval and U.S. government regulatory reviews, is estimated to be completed in the first quarter of 1998.

Forward-Looking Information

Certain statements and assumptions in Management's Discussion and Analysis contain or are based on "forward-looking" information (as defined in the Private Securities Litigation and Reform Act of 1995) that involves risk and uncertainties, including statements and assumptions with respect to future revenues, program performance and cash flows, the outcome of contingencies including litigation and environmental remediation, and anticipated costs of capital investments and planned dispositions. The company's operations are necessarily subject to various risks and uncertainties; actual outcomes are dependent upon many factors, including, without limitation, the company's successful performance of internal plans; government customers' budgetary restraints; customer changes in shortrange and long-range plans; domestic and international competition in both the defense and commercial areas; product performance; continued development and acceptance of new products; performance issues with key suppliers and subcontractors; government import and export policies; termination of government contracts; the outcome of political and legal processes; legal, financial, and governmental risks related to international transactions and global needs for military and commercial aircraft and electronic systems and support as well as other economic, political and technological risks and uncertainties.

Northrop Grumman Corporation and Subsidiaries

Part II OTHER INFORMATION

Item 1. Legal Proceedings

There have been no material changes in the litigation reported in the company's Form 10-K for the year ended December 31, 1996 except as reported in Item 1 of the company's Form 10-Q for the quarter ended March 31, 1997.

Northrop Grumman Corporation and Subsidiaries

Item 6. Exhibits and Reports on Form 8-K

(a) Exhibits

Exhibit 2 - Agreement and Plan of Merger, as amended as of September 29, 1997, by and among the Registrant, Lockheed Martin Corporation and Hurricane Sub, Inc. (Pursuant to item 601(b)(2) of Regulation S-K, the schedules thereto have been omitted but will be furnished supplementally to the Commission upon request.)

Exhibit 11 - Statement re Computation of Per Share Earnings

Exhibit 27 - Financial Data Schedule

(b) Reports on Form 8-K

A report on Form 8-K was filed with the Securities and Exchange Commission on July 9, 1997 regarding the combination of Northrop Grumman Corporation and Lockheed Martin Corporation.

A report on Form 8-K was filed with the Securities and Exchange Commission on August 29, 1997 regarding the acquisition of Logicon, Inc.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

Northrop Grumman Corporation (Registrant)

Date: November 7, 1997 by/s/N. F. Gibbs

Nelson F. Gibbs

Corporate Vice President and Controller

Date: November 7, 1997 by/s/James C. Johnson

James C. Johnson

Corporate Vice President and Secretary

EXHIBIT 11
STATEMENT RE COMPUTATION OF PER SHARE EARNINGS (in thousands, except per share)

	Three months ended September 30,			nths ended tember 30,
	1997	1996	1997	1996
Primary:				
Average shares outstanding	66 , 722	66 , 279	66,656	61,347
Common stock equivalents	1,305	1,643	1,065	1,627
Totals	68 , 027	67 , 922	67 , 721	62 , 974
Net income	\$97 , 705	\$77 , 593	\$290 , 089	\$239,413
Earnings per share(1)	\$ 1.44	\$ 1.14	\$ 4.28	\$ 3.80
Fully diluted:				
Average shares outstanding	66 , 722	66 , 279	66,656	61,347
Common stock equivalents	1,411	1,844	1,411	2,061
Totals	68,133	68,123	68 , 067	63,408
Net income	\$97 , 705	\$77 , 593	\$290 , 089	\$239,413
Earnings per share(1)	\$ 1.43	\$ 1.14	\$ 4.26	\$ 3.78

⁽¹⁾ This calculation was made in compliance with Item 601 of Regulation S-K. Earnings per share presented elsewhere in this report exclude from their calculation shares issuable under employee stock options and rights, since their dilutive effect is less than 3%.

```
9-MOS
     DEC-31-1997
         SEP-30-1997
                      48
              0
           1,008
84
1,315
3,093
3,230
1,850
9,778
             1,658
       2,670
                   3,020
          0
                   0
                    789
                 1,714
 9,778
            6,643
                    6,009
              6,009
             (30)
              0
            197
             467
           177
290
0
               0 0 290
               4.35
               4.35
```

AGREEMENT AND PLAN OF MERGER

by and among

LOCKHEED MARTIN CORPORATION,

HURRICANE SUB, INC.

and

NORTHROP GRUMMAN CORPORATION

Dated as of July 2, 1997

As Amended as of September 29, 1997

TABLE OF CONTENTS

		PAGE
THE MERG 1.1. 1.2. 1.3. 1.4. 1.5.	ER The Merger The Closing Effective Time Certificate of Incorporation and By-Laws Directors of the Surviving Corporation Officers of the Surviving Corporation	1 1 2 2 2 2
CONVERSI 2.1 2.2. 2.3.	ON AND EXCHANGE OF SECURITIES. Merger Sub Stock. Company Stock. Exchange of Certificates Representing Company Common Stock. Adjustment of Exchange Ratio.	2 2 2 3 5
3.1. 3.2. 3.3. 3.4. 3.5. 3.6. 3.7. 3.8. 3.9. 3.10. 3.11. 3.12. 3.13.	Existence; Good Standing; Corporate Authority; Compliance With Law. Authorization, Validity and Effect of Agreements. Capitalization. [Reserved]. No Violation. SEC Documents. Investigations; Litigation. Absence of Certain Changes. Taxes. Contracts. Employee Benefit Plans. No Brokers. Opinion of Financial Advisor.	5 5 6 6 7 7 7 8 8 8 8 9 9 10 10
3.15. 3.16. REPRESEN	Pooling of Interests; Tax Reorganization Environmental Matters TATIONS AND WARRANTIES OF PARENT	11 11 11 11 11 12 12 13 13 13 14
	1.1. 1.2. 1.3. 1.4. 1.5. 1.6. CONVERSI 2.1 2.2. 2.3. 2.4. REPRESEN 3.1. 3.2. 3.3. 3.4. 3.5. 3.6. 3.7. 3.8. 3.9. 3.10. 3.11. 3.12. 3.13. 3.14. 3.15. 3.16. REPRESEN AND ME 4.1. 4.2. 4.3. 4.4. 4.5. 4.6.	1.2. The Closing 1.3. Effective Time 1.4. Certificate of Incorporation and By-Laws 1.5. Directors of the Surviving Corporation 1.6. Officers of the Surviving Corporation. CONVERSION AND EXCHANGE OF SECURITIES 2.1 Merger Sub Stock 2.2. Company Stock. 2.3. Exchange of Certificates Representing

	4.10.	Contracts	14
	4.11.	Employee Benefit Plans	15
	4.12.	No Brokers	15
	4.13.	Opinion of Financial Advisor	16
	4.14.	Company Stock Ownership	16
	4.15.	Pooling of Interests; Tax Reorganization	16
	4.16.	Environmental Matters	16
	4.10.	Environmental Matters	Τ (
ARTICLE 5	COLLEGIANI	TI C	16
ARTICLE 3		IS	
	5.1.	Alternative Proposals	16
	5.2.	Interim Operations of the Company	17
	5.3.	Interim Operations of Parent	19
	5.4.	Meetings of Stockholders	20
	5.5.	Filings; Other Actions	20
	5.6.	HSR Act	20
	5.7.	Inspection of Records	21
	5.8.	Publicity	21
	5.9.	Registration Statement	21
	5.10.	Listing Application	22
	5.11.	Affiliate Letters	22
	5.12.		22
		Expenses	
	5.13.	Indemnity; Insurance	23
	5.14.	Employee Benefits	23
	5.15.	Reorganization	24
	5.16.	Shareholder Rights Plan	24
	5.17.	[Reserved]	24
	5.18.	Agreements	24
	5.19.	Parent Board of Directors	25
	5.20.	Takeover Statute	25
ARTICLE 6	CONDITIO	ONS	25
	6.1.	Conditions to Each Party's Obligation to	
	0.1.	Effect the Merger	25
	6.2.	Conditions to Obligation of the Company to	2.
	0.2.	· · · · · · · · · · · · · · · · · · ·	2.0
	6 0	Effect the Merger	26
	6.3.	Conditions to Obligation of Parent and	
		Merger Sub to Effect the Merger	26
ARTICLE 7		TION	27
	7.1.	Termination by Mutual Consent	27
	7.2.	Termination by Either Parent or the Company	27
	7.3.	Termination by the Company	28
	7.4.	Termination by Parent	28
	7.5.	Effect of Termination and Abandonment	28
	7.6.	Extension; Waiver	29
ARTICLE 8	GENERAL	PROVISIONS	29
	8.1.	Nonsurvival of Representations, Warranties	
		and Agreements	29
	8.2.	Notices	29
	8.3.	Assignment; Binding Effect	30
		=	

8.4.	Entire Agreement	30
8.5.	Amendment	31
8.6.	Governing Law	31
8.7.	Counterparts	31
8.8.	Headings	31
8.9.	Interpretation	31
8.10.	Waivers	31
8.11.	Incorporation of Exhibits	31
8.12.	Severability	32
8.13.	Enforcement of Agreement	32
8.14.	Subsidiaries	32

EXHIBITS A Form of Affiliate Letter

SCHEDULES

Company Disclosure Schedules

3.5.	No Violations
3.7.	Litigation
3.10.	Contracts
3.15.	Investigations; Litigation
3.10.	Contracts
3.11.	Employee Benefit Plans

3.15. Pooling of Interests 3.16. Environmental Matters
5.2. Interim Operations
5.14. Employee Benefit Agreements

Parent Disclosure Schedules

Litigation			
Contracts			
Employee Benefit	Plans	Investigations;	Litigation
Contracts			
Employee Benefit	Plans		
	Employee Benefit Contracts	Contracts Employee Benefit Plans	Contracts Employee Benefit Plans Investigations; Contracts

AGREEMENT AND PLAN OF MERGER

AGREEMENT AND PLAN OF MERGER (this "Agreement"), dated as of July 2, 1997, between Lockheed Martin Corporation, a Maryland corporation ("Parent"), Hurricane Sub, Inc., a Delaware corporation and a wholly owned subsidiary of Parent ("Merger Sub"), and Northrop Grumman Corporation, a Delaware corporation (the "Company").

RECTTALS

The Boards of Directors of Parent and the Company each have determined that a business combination between Parent and the Company is in the best interests of their respective companies and stockholders and presents an opportunity for their respective companies to achieve long-term strategic and financial benefits, and accordingly have agreed to effect the merger provided for herein upon the terms and subject to the conditions set forth herein.

It is intended that for federal income tax purposes, the merger provided for herein shall qualify as a reorganization within the meaning of Section 368 of the Internal Revenue Code of 1986, as amended (the "Code"), and that for financial accounting purposes the Merger shall be accounted for as a pooling of interests.

Merger Sub is a wholly owned subsidiary of Parent and has been formed solely to facilitate the Merger (as defined herein) and will conduct no business or activity other than in connection with the Merger.

NOW, THEREFORE, in consideration of the foregoing, and of the representations, warranties, covenants and agreements contained herein, the parties hereto hereby agree as follows:

ARTICLE 1

THE MERGER

- 1.1. The Merger. Subject to the terms and conditions of this Agreement, at the Effective Time (as defined in Section 1.3), Merger Sub shall be merged with and into the Company in accordance with this Agreement, and the separate corporate existence of Merger Sub shall thereupon cease (the "Merger"). The Company shall be the surviving corporation in the Merger (sometimes hereinafter referred to as the "Surviving Corporation") and will be a wholly owned subsidiary of Parent. The Merger shall have the effects specified in the Delaware General Corporation Law (the "DGCL").
- 1.2. The Closing. Subject to the terms and conditions of this Agreement, the closing of the Merger (the "Closing") shall take place (a) at the offices of Dewey Ballantine, 1301 Avenue of the Americas, New York, New York 10019, at 10:00 a.m., local time, on the first business day immediately following the day on which the last to be fulfilled or waived of the conditions set forth in Article 6 shall be fulfilled or waived in accordance herewith or (b) at such other time, date or place as Parent and the Company may agree. The date on which the Closing occurs is hereinafter referred to as the "Closing Date."

- 1.3. Effective Time. If all the conditions set forth in Article 6 shall have been fulfilled or waived in accordance herewith and this Agreement shall not have been terminated as provided in Article 7, the parties hereto shall cause a Certificate of Merger meeting the requirements of Section 251 of the DGCL to be properly executed and filed in accordance with such Section on the Closing Date. The Merger shall become effective at the time of filing of the Certificate of Merger with the Secretary of State of the State of Delaware in accordance with the DGCL or at such later time which the parties hereto shall have agreed upon and designated in such filing as the effective time of the Merger (the "Effective Time").
- 1.4. Certificate of Incorporation and By-laws. The Certificate of Incorporation and By-Laws of Merger Sub in effect immediately prior to the Effective Time shall be the Certificate of Incorporation and By-Laws of the Surviving Corporation, until duly amended in accordance with applicable law.
- 1.5. Directors of the Surviving Corporation. The directors of Merger Sub immediately prior to the Effective Time shall be the directors of the Surviving Corporation as of the Effective Time and until their successors are duly appointed or elected in accordance with applicable law.
- 1.6. Officers of the Surviving Corporation. The officers of the Company immediately prior to the Effective Time shall be the officers of the Surviving Corporation as of the Effective Time and until their successors are duly appointed or elected in accordance with applicable law.

ARTICLE 2

CONVERSION AND EXCHANGE OF SECURITIES

- 2.1. Merger Sub Stock. At the Effective Time, each share of common stock, par value \$.01 per share, of Merger Sub outstanding immediately prior to the Effective Time shall be converted into and exchanged for one validly issued, fully paidand non- and non- non-assessable share of common stock, par value \$1.00 per share, of the Surviving Corporation.
- 2.2. Company Stock. (a) At the Effective Time, each share of common stock, par value \$1.00 per share (the "Company Common Stock"), of the Company issued and outstanding immediately prior to the Effective Time shall, by virtue of the Merger and without any action on the part of the holder thereof, be converted into the right to receive 1.1923 shares of common stock, par value \$1.00 per share (the "Parent Common Stock"), of Parent (the "Exchange Ratio").
- (b) As a result of the Merger and without any action on the part of the holder thereof, at the Effective Time all shares of Company Common Stock shall cease to be outstanding and shall be cancelled and retired and shall cease to exist, and each holder of shares of Company Common Stock shall thereafter cease to have any rights with respect to such shares of Company Common Stock, except the right to receive, without interest, the Exchange Ratio and cash for

fractional shares of Parent Common Stock in accordance with Sections 2.3(b) and 2.3(e) upon the surrender of a certificate (a "Certificate") representing such shares of Company Common Stock.

- (c) Each share of Company Common Stock issued and held in the Company's treasury at the Effective Time shall, by virtue of the Merger, cease to be outstanding and shall be cancelled and retired without payment of any consideration therefor.
- (d) At the Effective Time, each option to purchase shares of Company Common Stock outstanding at the Effective Time under any Company stock option plan (a "Company Option") shall, by virtue of the Merger and without any further action on the part of the Company or the holder of any such Company Option, be assumed by Parent and shall be converted into an option to purchase Parent Common Stock. Each Company Option assumed by Parent shall be exercisable upon the same terms and conditions as under the applicable Company stock option plan and the applicable option agreement issued thereunder, except that (i) each such Company Option shall be exercisable for that whole number of shares of Parent Common Stock (rounded to the nearest whole share) into which the number of shares of Company Common Stock subject to such Company Option immediately prior to the Effective Time would be converted under Section 2.2(a), and (ii) the option price per share of Parent Common Stock shall be an amount equal to the option price per share of Company Common Stock subject to such Company Option divided by the Exchange Ratio (the option price per share being rounded to the nearest full cent). No payment shall be made for fractional interests, rather, the aggregate number of shares to be issued under any assumed Company Option shall be rounded to the nearest whole number. At the Effective Time, each share of Company Common Stock subject to transfer or vesting restrictions (the "Restricted Stock") shall upon conversion into the Parent Common Stock under Section 2.2(a) be subject to the same terms and conditions, including transfer restrictions and vesting schedule, as the Restricted Stock. The number of shares of Company Common Stock issued as of the Effective Time for the restricted performance stock rights shall be determined in accordance with the provisions of the Company's 1993 Long Term Incentive Plan and shall be converted into Parent Common Stock under Section 2.2(a).
 - 2.3. Exchange of Certificates Representing Company Common Stock.
- (a) As of the Effective Time, Parent shall deposit, or shall cause to be deposited, with an exchange agent selected by Parent (the "Exchange Agent"), for the benefit of the holders of shares of Company Common Stock, for exchange in accordance with this Article 2, certificates representing the shares of Parent Common Stock to be issued in connection with the Merger and Parent's good faith estimate of the cash in lieu of fractional shares expected to be payable in connection with the Merger (such cash and certificates for shares of Parent Common Stock, together with any dividends or distributions with respect thereto (relating to record dates for such dividends or distributions after the Effective Time), being hereinafter referred to as the "Exchange Fund") to be issued pursuant to Section 2.2 and paid pursuant to this Section 2.3 in exchange for outstanding shares of Company Common Stock.
- (b) Promptly after the Effective Time, Parent shall cause the Exchange Agent to mail to each holder of record of shares of Company Common Stock (i) a letter of transmittal which shall specify that delivery shall be effected, and risk of loss and title to such shares of Company Common Stock shall pass, only upon delivery of the Certificates representing such shares to the

Exchange Agent and which shall be in such form and have such other provisions as Parent may reasonably specify and (ii) instructions for use in effecting the surrender of such Certificates in exchange for the Exchange Ratio and cash in lieu of fractional shares. Upon surrender of a Certificate for cancellation to the Exchange Agent together with such letter of transmittal, duly executed and completed in accordance with the instructions thereto, the holder of the shares represented by such Certificate shall be entitled to receive in exchange therefor (x) a certificate representing that number of whole shares of Parent Common Stock and (y) a check representing the amount of cash in lieu of fractional shares, if any, and unpaid dividends and distributions, if any which such holder has the right to receive in respect of the Certificate surrendered pursuant to the provisions of this Article 2, after giving effect to any required withholding tax, and the shares represented by the Certificate so surrendered shall forthwith be cancelled. No interest will be paid or accrued on the cash payable to holders of shares of Company Common Stock. In the event of a transfer of ownership of Company Common Stock which is not registered in the transfer records of the Company, a certificate representing the proper number of shares of Parent Common Stock, together with a check for the cash to be paid may be issued to such a transferee if the Certificate representing such Company Common Stock is presented to the Exchange Agent, accompanied by all documents required to evidence and effect such transfer and to evidence that any applicable stock transfer taxes have been paid.

- (c) Notwithstanding any other provisions of this Agreement, no dividends or other distributions declared after the Effective Time on Parent Common Stock shall be paid with respect to any shares of Company Common Stock represented by a Certificate until such Certificate is surrendered for exchange as provided herein. Subject to the effect of applicable laws, following surrender of any such Certificate, there shall be paid to the holder of the certificates representing whole shares of Parent Common Stock issued in exchange therefor, without interest, (i) at the time of such surrender, the amount of dividends or other distributions with a record date after the Effective Time theretofore payable with respect to such whole shares of Parent Common Stock and not paid, less the amount of any withholding taxes which may be required thereon, and (ii) at the appropriate payment date, the amount of dividends or other distributions with a record date after the Effective Time but prior to surrender and a payment date subsequent to surrender payable with respect to such whole shares of Parent Common Stock, less the amount of any withholding taxes which may be required thereon.
- (d) At or after the Effective Time, there shall be no transfers on the stock transfer books of the Company of the shares of Company Common Stock which were outstanding immediately prior to the Effective Time. If, after the Effective Time, Certificates are presented to the Surviving Corporation, they shall be cancelled and exchanged for certificates for shares of Parent Common Stock and cash deliverable in respect thereof pursuant to this Agreement in accordance with the procedures set forth in this Article 2. Certificates surrendered for exchange by any person constituting an "affiliate" of the Company for purposes of Rule 145(c) under the Securities Act of 1933, as amended (the "Securities Act"), shall not be exchanged until Parent has received a written agreement from such person as provided in Section 5.11.
- (e) No fractional shares of Parent Common Stock shall be issued pursuant hereto. In lieu of the issuance of any fractional share of Parent Common Stock, cash adjustments will be paid to holders in respect of any fractional share of Parent Common Stock that would otherwise be issuable, and the amount of such cash adjustment shall be equal to such fractional

proportion of the "Average Price" of a share of Parent Common Stock. The "Average Price" of a share of Parent Common Stock shall be the average of the closing sales prices thereof as reported on The New York Stock Exchange (the "NYSE") Composite Tape (as reported by The Wall Street Journal or, if not reported thereby, by another authoritative source) over the ten (10) business days immediately preceding the Closing Date.

- (f) Any portion of the Exchange Fund (including the proceeds of any investments thereof and any shares of Parent Common Stock) that remains unclaimed by the former stockholders of the Company one year after the Effective Time shall be delivered to the Surviving Corporation. Any former stockholders of the Company who have not theretofore complied with this Article 2 shall thereafter look only to the Surviving Corporation for payment of their shares of Parent Common Stock, cash and unpaid dividends and distributions on Parent Common Stock deliverable in respect of each share of Company Common Stock such stockholder holds as determined pursuant to this Agreement, in each case, without any interest thereon.
- (g) None of Parent, the Company, the Surviving Corporation, the Exchange Agent or any other person shall be liable to any former holder of shares of Company Common Stock for any amount properly delivered to a public official pursuant to applicable abandoned property, escheat or similar laws.
- (h) In the event any Certificate shall have been lost, stolen or destroyed, upon the making of an affidavit of that fact by the person claiming such Certificate to be lost, stolen or destroyed and, if required by the Surviving Corporation, the posting by such person of a bond in such reasonable amount as the Surviving Corporation may direct as indemnity against any claim that may be made against it with respect to such Certificate, the Exchange Agent will issue in exchange for such lost, stolen or destroyed Certificate the shares of Parent Common Stock and cash deliverable in respect thereof pursuant to this Agreement.

2.4. Adjustment of Exchange Ratio.

In the event that, subsequent to the date of this Agreement but prior to the Effective Time, the outstanding shares of Parent Common Stock or Company Common Stock, respectively, shall have been changed into a different number of shares or a different class as a result of a stock split, reverse stock split, stock dividend, subdivision, reclassification, combination, exchange, recapitalization or other similar transaction, the Exchange Ratio shall be appropriately adjusted.

ARTICLE 3

REPRESENTATIONS AND WARRANTIES OF THE COMPANY

Except as set forth in the disclosure schedule delivered at or prior to the execution hereof to Parent (the "Company Disclosure Schedule") or in the Company Reports (as defined below), the Company represents and warrants to Parent as of the date of this Agreement as follows:

3.1. Existence; Good Standing; Corporate Authority; Compliance With Law. The Company is a corporation duly incorporated, validly existing and in good standing under the laws of its jurisdiction of incorporation. The Company is duly licensed or qualified to do business

as a foreign corporation and is in good standing under the laws of any other state of the United States in which the character of the properties owned or leased by it or in which the transaction of its business makes such qualification necessary, except where the failure to be so qualified or to be in good standing would not have a Company Material Adverse Effect. For the purposes of this Agreement, "Company Material Adverse Effect" means a material adverse effect on the business, results of operations or financial condition of the Company and its Subsidiaries (as defined in Section 8.14) taken as a whole, other than any effects or changes arising out of, resulting from or relating to general economic, financial or industry conditions. The Company has all requisite corporate power and authority to own, operate and lease its properties and carry on its business as now conducted. Each of the Company's Significant Subsidiaries (as defined in Section 8.14 hereof) is a corporation or partnership duly organized, validly existing and in good standing under the laws of its jurisdiction of incorporation or organization, has the corporate or partnership power and authority to own its properties and to carry on its business as it is now being conducted, and is duly qualified to do business and is in good standing in each jurisdiction in which the ownership of its property or the conduct of its business requires such qualification, except for jurisdictions in which such failure to be so qualified or to be in good standing would not have a Company Material Adverse Effect. Neither the Company nor any of its Subsidiaries is in violation of any order of any court, governmental authority or arbitration board or tribunal, or any law, ordinance, governmental rule or regulation to which the Company or any of its Subsidiaries or any of their respective properties or assets is subject, other than any violations which would not have a Company Material Adverse Effect. The Company and its Subsidiaries have obtained all licenses, permits and other authorizations and have taken all actions required by applicable law or governmental regulations in connection with their business as now conducted, except where the failure to obtain any such item or to take any such action would not have a Company Material Adverse Effect.

- 3.2. Authorization, Validity and Effect of Agreements. The Company has the requisite corporate power and authority to execute and deliver this Agreement and all agreements and documents contemplated hereby. Subject only to the approval of this Agreement and the transactions contemplated hereby by the holders of a majority of the outstanding shares of Company Common Stock, the consummation by the Company of the transactions contemplated hereby has been approved by the Board of Directors of the Company and duly authorized by all requisite corporate action. This Agreement constitutes, and all agreements and documents contemplated hereby (when executed and delivered pursuant hereto for value received) will constitute, the valid and legally binding obligations of the Company, enforceable in accordance with their respective terms, subject to applicable bankruptcy, insolvency, moratorium or other similar laws relating to creditors' rights and general principles of equity.
- 3.3. Capitalization. The authorized capital stock of the Company consists of 200,000,000 shares of Company Common Stock and 10,000,000 shares of preferred stock, par value \$1.00 per share (the "Company Preferred Stock"). As of May 31, 1997, there were 58,050,08558,059,010 shares of Company Common Stock, and no shares of Company Preferred Stock, issued and outstanding. From such date to the date of this Agreement, no additional shares of capital stock of the Company have been issued, except pursuant to the exercise of options outstanding under Company Stock Option Plans. As of May 31, 1997, options to acquire 3,505,996 shares of Company Common Stock were outstanding. From such date to the date of this Agreement, no additional options have been granted. The Company has no outstanding bonds, debentures, notes or other obligations the holders of which have the right to vote (or which are convertible into or exercisable for securities having the right to vote) with the stockholders of the Company on any matter. All issued and outstanding shares of Company Common Stock are duly authorized, validly issued, fully paid, nonassessable and free of preemptive rights. Except as described above, and except for 1,818,143 restricted performance stock rights, 40,290 restricted stock rights, and 5,750 restricted award shares of the Company outstanding as of the date hereof, there are not at the date of this Agreement any existing options, warrants, calls, subscriptions, convertible securities, or other rights, agreements or commitments (other than the Agreement and Plan of Merger, dated May 4, 1997, between the Company, Logicon, Inc. and NG Acquisition, Inc. (the "Logicon Merger Agreement")) which obligate the Company or any of its Subsidiaries to issue, transfer or sell any shares of capital stock of the Company or any of its Subsidiaries. After the Effective Time, the Surviving Corporation will have no obligation to issue, transfer or sell any shares of capital stock or other securities of the Company or the Surviving Corporation pursuant to any Company Benefit Plan (as defined in Section 3.11).

3.4. [Reserved].

- 3.5. No Violation. Except as set forth in Section 3.5 of the Company Disclosure Schedule, neither the execution and delivery by the Company of this Agreement nor the consummation by the Company of the transactions contemplated hereby in accordance with the terms hereof, will: (i) conflict with or result in a breach of any provisions of the Certificate of Incorporation or Bylaws of the Company; (ii) result in a breach or violation of, a default under, or the triggering of any payment or other material obligations pursuant to, or accelerate vesting under, any existing Company Stock Option Plan, or any grant or award made under any of the foregoing; (iii) violate, conflict with, result in a breach of any provision of, constitute a default (or an event which, with notice or lapse of time or both, would constitute a default) under, result in the termination or in a right of termination or cancellation of, accelerate the performance required by, result in the triggering of any payment or other material obligations pursuant to, result in the creation of any lien, security interest, charge or encumbrance upon any of the material properties of the Company or its Subsidiaries under, or result in being declared void, voidable, or without further binding effect, any of the terms, conditions or provisions of any note, bond, mortgage, indenture, deed of trust or any material license, franchise, permit, lease, contract, agreement or other instrument, commitment or obligation to which the Company or any of its Subsidiaries is a party, or by which the Company or any of its Subsidiaries or any of their respective properties is bound or affected, except for any of the foregoing matters which would not have a Company Material Adverse Effect; or (iv) other than the filings provided for in Article 1, applicable federal, state and local regulatory filings, filings required under the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended (the "HSR Act"), the Securities Exchange Act of 1934, as amended (the "Exchange Act"), the Securities Act of 1933, as amended (the "Securities Act"), or applicable state securities and "Blue Sky" laws or filings in connection with the maintenance of qualification to do business in other jurisdictions (collectively, the "Regulatory Filings"), require any consent, approval or authorization of, or declaration, filing or registration with, any domestic governmental or regulatory authority, other than approvals or filings which, if not obtained or made would not have a Company Material Adverse Effect.
- 3.6. SEC Documents. For the purposes of this Agreement, the "Company Reports" means each registration statement, report, proxy statement or information statement (as defined in Regulation 14C under the Exchange Act) of the Company prepared by it since January 1,

1995, in the form (including exhibits and any amendments thereto) filed with the Securities and Exchange Commission (the "SEC"). As of their respective dates, the Company Reports (i) complied as to form in all material respects with the applicable requirements of the Securities Act, the Exchange Act, and the rules and regulations thereunder and (ii) did not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements made therein, in the light of the circumstances under which they were made, not misleading. Each of the consolidated balance sheets of the Company included in or incorporated by reference into the Company Reports (including the related notes and schedules) fairly presents the consolidated financial position of the Company and its Subsidiaries as of its date, and each of the consolidated statements of income, retained earnings and cash flows of the Company included in or incorporated by reference into the Company Reports (including any related notes and schedules) fairly presents the results of operations, retained earnings or cash flows, as the case may be, of the Company and its Subsidiaries for the periods set forth therein (subject, in the case of unaudited statements, to normal year-end audit adjustments which would not be material in amount or effect), in each case in accordance with generally accepted accounting principles consistently applied during the periods involved, except as may be noted therein. Neither the Company nor any of its Subsidiaries has any liabilities or obligations required to be disclosed in a balance sheet or the notes thereto prepared in accordance with generally accepted accounting principles consistently applied, except (a) liabilities or obligations reflected on, or reserved against in, a balance sheet of the Company or in the notes thereto, and included in the Company Reports and (b) liabilities or obligations incurred since March 31, 1997 in the ordinary course of business.

- 3.7. Investigations; Litigation. Except as set forth in Section 3.7 or Section 3.10 of the Company Disclosure Schedule, (a) to the knowledge of the Company, no material investigation by any governmental entity with respect to the Company or any of its Subsidiaries is pending or threatened nor has any governmental entity indicated to the Company an intention to conduct the same except for any of the foregoing which would not have a Company Material Adverse Effect; and (b) there are no actions, suits or proceedings pending against the Company or its Subsidiaries or, to the knowledge of the Company, threatened against the Company or its Subsidiaries, at law or in equity, or before or by any federal or state commission, board, bureau, agency or instrumentality, that are reasonably likely to have a Company Material Adverse Effect.
- 3.8. Absence of Certain Changes. Since December 31, 1996, the Company has conducted its business only in the ordinary course of such business, and there has not been (i) any Company Material Adverse Effect or any event which is reasonably likely to result in a Company Material Adverse Effect; (ii) any declaration, setting aside or payment of any dividend or other distribution with respect to its capital stock (other than the regular quarterly cash dividends of \$0.40 per share, payable on Company Common Stock); or (iii) any material change in its accounting principles, practices or methods.
- 3.9. Taxes. The Company and each of its Subsidiaries (i) have timely filed all material federal, state and foreign tax returns required to be filed by any of them for tax years ended prior to the date of this Agreement or requests for extensions have been timely filed and any such request shall have been granted and not expired, and all such returns are complete in all material respects, (ii) have paid or accrued all taxes shown to be due and payable on such returns

8

and (iii) have properly accrued all such taxes for such periods subsequent to the periods covered by such returns.

- 3.10. Contracts. (a) Except as disclosed in Section 3.7 or Section 3.10 of the Company Disclosure Schedule, to the knowledge of the Company, with respect to Government Contracts, there is, as of the date hereof, no (i) civil fraud or criminal investigation by any government investigative agency that is reasonably likely to have a Company Material Adverse Effect, (ii) suspension or debarment proceeding (or equivalent proceeding) against the Company or any of its Subsidiaries that is reasonably likely to have a Company Material Adverse Effect, (iii) request by the government for a contract price adjustment based on a claimed disallowance by Defense Contract Audit Agency or claim of defective pricing in excess of \$40 million, (iv) dispute between the Company or any of its Subsidiaries and the U.S. Government which, since June 30, 1996, has resulted in a government contracting officer's determination and finding final decision where the amount in controversy exceeds or is expected to exceed \$40 million or (v) claim or equitable adjustment by the Company or any of its Subsidiaries against the U.S. Government or any third party in excess of \$40 million.
- (b) For the purposes of this Agreement, with respect to any party, "Government Contract" means any prime contract, subcontract, teaming agreement or arrangement, joint venture, basic ordering agreement, pricing agreement, letter contract, purchase order, delivery order, change order, Bid or other arrangement of any kind between such party or any of its Subsidiaries and (i) the U.S. Government (acting on its own behalf or on behalf of another country or international organization), (ii) any prime contractor of the U.S. Government or (iii) any subcontractor with respect to any contract of a type described in clauses (i) or (ii) above. For the purposes of this Agreement, with respect to any party, "Bid" means any quotation, bid or proposal made by such party or any of its Subsidiaries that if accepted or awarded would lead to a Contract with the U.S. Government or any other person for the design, manufacture and sale of products or the provision of services. For the purposes of this Agreement, with respect to any party, "Contracts" means all contracts, agreements, leases (including leases of real property), licenses, commitments, sales and purchase orders, intercompany work transfer agreements) with respect to work by or for another of such party's businesses) and other instruments of any kind, whether written or oral.

3.11. Employee Benefit Plans.

For the purpose of this Agreement, "Company Benefit Plans" means all employee benefit plans and other benefit arrangements covering employees or former employees of the Company and its Subsidiaries and all employee agreements providing compensation, severance or other benefits to any employee or former employee of the Company or any of its Subsidiaries. With respect to each Company Benefit Plan that is intended to be a "qualified plan" within the meaning of Section 401(a) of the Internal Revenue Code (the "Code"), either the Internal Revenue Service (the "IRS") has issued a favorable determination letter that has not been revoked, or an application for a favorable determination letter was timely submitted to the IRS for which no final action has been taken by the IRS. To the knowledge of the Company, there is no reason that is not susceptible to cure why the qualified status under Section 401(a) of the Code of any Company Benefit Plan would be denied or revoked, whether retroactively or prospectively. Except as would not have a Company Material Adverse Effect, no Company Benefit Plan, any fiduciary thereof, nor the Company has incurred any liability or penalty under Section 4975 of the Code or Section 502(i)

of ERISA. Except as would not have a Company Material Adverse Effect, each Company Benefit Plan has been maintained and administered in all material respects in compliance with its terms and with the Employee Retirement Income Security Act of 1974, as amended ("ERISA"), and the Code to the extent applicable thereto. Except as would not have a Company Material Adverse Effect, neither the Company nor any ERISA Affiliate (during the period of its affiliated status and prior thereto, to its knowledge) has any existing liability currently due and payable that has not been satisfied in full under Title IV of ERISA or Section 412 of the Code. To the knowledge of the Company, there are no current plans to terminate, whether voluntarily or involuntarily any materially underfunded pension plans of the Company or any ERISA Affiliate that are subject to Title IV of ERISA. Except as would not have a Company Material Adverse Effect, to the knowledge of the Company, there are no pending or anticipated claims against or otherwise involving any of the Company Benefit Plans and no suit, action or other litigation (excluding claims for benefits incurred in the ordinary course of the Company Benefit Plan activities) has been brought against or with respect to any such Company Benefit Plan, except for any of the foregoing which would not have a Company Material Adverse Effect. All material contributions required to be made as of the date hereof to the Company Benefit Plans have been made or provided for. Except as set forth in Section 3.11 of the Company Disclosure Schedule, the execution of, and performance of the transactions contemplated in, this Agreement will not (either alone or upon the occurrence of any additional or subsequent events) constitute an event under any benefit plan, policy, arrangement or agreement or any trust or loan that will or may result in any payment (whether of severance pay or otherwise), acceleration, forgiveness of indebtedness, vesting, distribution, increase in benefits or obligation to fund benefits with respect to any employee. The only material severance agreements or severance policies applicable to the Company or its Subsidiaries are the agreements and policies specifically referred to in Section 3.11 of the Company Disclosure Schedule.

For purposes of this Agreement "ERISA Affiliate" means any business or entity which is a member of the same "controlled group of corporations," under "common control" or an "affiliated service group" with an entity within the meanings of Sections 414(b), (c) or (m) of the Code, or required to be aggregated with the entity under Section 414(o) of the Code, or is under "common control" with the entity, within the meaning of Section 4001(a)(14) of ERISA, or any regulations promulgated or proposed under any of the foregoing Sections.

- 3.12. No Brokers. The Company has not entered into any contract, arrangement or understanding with any person or firm which may result in the obligation of the Company or Parent to pay any finder's fees, brokerage or agent's commissions or other like payments in connection with the negotiations leading to this Agreement or the consummation of the transactions contemplated hereby, except that the Company has retained Salomon Brothers Inc as its financial advisor, the arrangements with which have been disclosed in writing to Parent prior to the date hereof. Other than the foregoing arrangements, the Company is not aware of any claim for payment of any finder's fees, brokerage or agent's commissions or other like payments in connection with the negotiations leading to this Agreement or the consummation of the transactions contemplated hereby.
- 3.13. Opinion of Financial Advisor. The Company has received the opinion of Salomon Brothers Inc to the effect that, as of the date hereof, the Exchange Ratio is fair to the holders of Company Common Stock from a financial point of view.

- 3.14. Parent Stock Ownership. Neither the Company nor any of its Subsidiaries owns any shares of Parent Common Stock or other securities convertible into Parent Common Stock.
- 3.15. Pooling of Interests; Tax Reorganization. In the judgment of the Company, the Company and its Subsidiaries have not taken (or as of the date hereof failed to take) any action which would prevent the accounting for the Merger as a pooling of interests in accordance with Accounting Principles Board Opinion No. 16 ("APB No. 16"), the interpretative releases issued pursuant thereto, and the pronouncements of the SEC. To the knowledge of the Company, the Company has not taken or failed to take any action which would prevent the Merger from constituting a reorganization within the meaning of section 368(a) of the Code.
- 3.16. Environmental Matters. Except as set forth in Section 3.16 of the Company Disclosure Schedule or as described in the Company Reports, the Company and each of its Subsidiaries are in material compliance with all applicable federal, state and local laws and regulations relating to pollution or protection of human health or the environment (including, without limitation, ambient air, surface water, ground water, land surface or subsurface strata) (collectively, "Environmental Laws"), except for instances of noncompliance that individually or in the aggregate do not, and, insofar as reasonably can be foreseen, in the future would not, have a Company Material Adverse Effect. Such compliance includes, but is not limited to, the possession by the Company and its Subsidiaries of all material permits and other governmental authorizations required under applicable Environmental Laws, and compliance with the terms and conditions thereof. Neither the Company nor any of its Subsidiaries has received written notice of, or to the knowledge of the Company, is the subject of, any actions, causes of action, claims, investigations, demands or notices by any person or entity alleging liability under or noncompliance with any Environmental Law ("Environmental Claims") that individually or in the aggregate would have a Company Material Adverse Effect. To the knowledge of the Company, there are no circumstances that are reasonably likely to prevent or interfere with such material compliance in the future.

ARTICLE 4

REPRESENTATIONS AND WARRANTIES OF PARENT AND MERGER SUB

Except as set forth in the disclosure schedule delivered at or prior to the execution hereof to the Company (the "Parent Disclosure Schedule") or in the Parent Reports (as defined below), Parent and Merger Sub represent and warrant to the Company as of the date of this Agreement as follows:

4.1. Existence; Good Standing; Corporate Authority; Compliance With Law. Each of Parent and Merger Sub is a corporation duly incorporated, validly existing and in good standing under the laws of its jurisdiction of incorporation. Parent is duly licensed or qualified to do business as a foreign corporation and is in good standing under the laws of any other state of the United States in which the character of the properties owned or leased by it or in which the transaction of its business makes such qualification necessary, except where the failure to be so qualified or to be in good standing would not have a Parent Material Adverse Effect. For the purposes of this Agreement, "Parent Material Adverse Effect" means a material adverse effect on the business, results of operations or financial condition of Parent and its Subsidiaries taken as a

whole, other than effects or changes arising out of, resulting from or relating to general economic, financial or industry conditions. Parent has all requisite corporate power and authority to own, operate and lease its properties and carry on its business as now conducted. Each of Parent's Significant Subsidiaries is a corporation or partnership duly organized, validly existing and in good standing under the laws of its jurisdiction of incorporation or organization, has the corporate or partnership power and authority to own its properties and to carry on its business as it is now being conducted, and is duly qualified to do business and is in good standing in each jurisdiction in which the ownership of its property or the conduct of its business requires such qualification, except for jurisdictions in which such failure to be so qualified or to be in good standing would not have a Parent Material Adverse Effect. Neither Parent nor any Parent Subsidiary is in violation of any order of any court, governmental authority or arbitration board or tribunal, or any law, ordinance, governmental rule or regulation to which Parent or any of its Subsidiaries or any of their respective properties or assets is subject, other than any violations which would not have a Parent Material Adverse Effect. Parent and its Subsidiaries have obtained all licenses, permits and other authorizations and have taken all actions required by applicable law or governmental regulations in connection with their business as now conducted, except where the failure to obtain any such item or to take any such action would not have a Parent Material Adverse Effect.

- 4.2. Authorization, Validity and Effect of Agreements. Each of Parent and Merger Sub has the requisite corporate power and authority to execute and deliver this Agreement and all agreements and documents contemplated hereby. Subject only to the approval of the issuance of Parent Common Stock pursuant to this Agreement by the holders of a majority of the outstanding shares of Parent Common Stock (the "Parent Stockholder Approval"), the consummation by Parent and Merger Sub of the transactions contemplated hereby has been approved by the Board of Directors of Parent and duly authorized by all requisite corporate action. This Agreement constitutes, and all agreements and documents contemplated hereby (when executed and delivered pursuant hereto for value received) will constitute, the valid and legally binding obligations of Parent and Merger Sub, enforceable in accordance with their respective terms, subject to applicable bankruptcy, insolvency, moratorium or other similar laws relating to creditors' rights and general principles of equity.
- 4.3. Capitalization. The authorized capital stock of Parent consists of 750,000,000 shares of Parent Common Stock, 50,000,000 shares of Series Preferred Stock, par value \$1.00 per share (the "Parent Series Preferred Stock") and 20,000,000 shares of Series A preferred stock, par value \$1.00 per share (the "Parent Preferred Stock"). As of April 30, 1997, there were 193,128,187 shares of Parent Common Stock and 20,000,000 shares of Parent Preferred Stock issued and outstanding. From such date to the date of this Agreement, no additional shares of capital stock of Parent have been issued, except pursuant to the exercise of options outstanding under Parent's stock option and employee stock purchase plans (the "Parent Stock Option Plans"). As of May 30, 1997, options to acquire 11,833,826 shares of Parent Common Stock were outstanding. date to the date of this Agreement, no additional options have been granted. Parent has no outstanding bonds, debentures, notes or other obligations the holders of which have the right to vote (or which are convertible into or exercisable for securities having the right to vote) with the stockholders of Parent on any matter. All such issued and outstanding shares of Parent Common Stock are duly authorized, validly issued, fully paid, nonassessable and free of preemptive rights. Except as contemplated by this Agreement, there are not at the date of this Agreement any existing options, warrants, calls, subscriptions, convertible securities, or other

12

rights, agreements or commitments which obligate Parent or any of its Subsidiaries to issue, transfer or sell any shares of capital stock of Parent or any of its Subsidiaries (other than under Parent Stock Option Plans).

- 4.4. Merger Sub. The authorized capital stock of Merger Sub consists of 100 shares of common stock, par value \$.01 per share, all of which shares are issued and outstanding and owned by Parent. Notwithstanding any provisions to the contrary, Parent may, in its sole discretion, increase the number of shares of authorized common stock of Merger Sub and the number of shares of common stock of Merger Sub issued and outstanding owned by Parent. Merger Sub has not engaged in any activities other than in connection with the transactions contemplated by this Agreement.
- 4.5. No Violation. Neither the execution and delivery by Parent and Merger Sub of this Agreement, nor the consummation by Parent and Merger Sub of the transactions contemplated hereby in accordance with the terms hereof, will: (i) conflict with or result in a breach of any provisions of the Certificate of Incorporation or Bylaws of Parent or Merger Sub; (ii) result in a breach or violation of, a default under, or the triggering of any payment or other material obligations pursuant to, or accelerate vesting under, any of the Parent Stock Option Plans, or any grant or award under any of the foregoing; (iii) violate, conflict with, result in a breach of any provision of, constitute a default (or an event which, with notice or lapse of time or both, would constitute a default) under, result in the termination or in a right of termination or cancellation of, accelerate the performance required by, result in the triggering of any payment or other material obligations pursuant to, result in the creation of any lien, security interest, charge or encumbrance upon any of the material properties of Parent or its Subsidiaries under, or result in being declared void, voidable, or without further binding effect, any of the terms, conditions or provisions of any note, bond, mortgage, indenture, deed of trust or any material license, franchise, permit, lease, contract, agreement or other instrument, commitment or obligation to which Parent or any of its Subsidiaries is a party, or by which Parent or any of its Subsidiaries or any of their respective properties is bound or affected, except for any of the foregoing matters which would not have a Parent Material Adverse Effect; or (iv) other than the Regulatory Filings, require any consent, approval or authorization of, or declaration, filing or registration with, any domestic governmental or regulatory authority, other than approvals or filings which, if not obtained or made would not have a Parent Material Adverse Effect.
- 4.6. SEC Documents. For the purposes of this Agreement, the "Parent Reports" means each registration statement, report, proxy statement or information statement of Parent prepared by it since January 1, 1995, in the form (including exhibits and any amendments thereto) filed with the SEC. As of the respective dates, the Parent Reports (i) complied as to form in all material respects with the applicable requirements of the Securities Act, the Exchange Act, and the rules and regulations thereunder and (ii) did not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements made therein, in the light of the circumstances under which they were made, not misleading. Each of the consolidated balance sheets included in or incorporated by reference into the Parent Reports (including the related notes and schedules) fairly presents the consolidated financial position of Parent and its Subsidiaries as of its date, and each of the consolidated statements of income, retained earnings and cash flows included in or incorporated by reference into the Parent Reports (including any related notes and schedules) fairly presents the results of operations, retained

earnings or cash flows, as the case may be, of Parent and its Subsidiaries for the periods set forth therein (subject, in the case of unaudited statements, to normal year-end audit adjustments which would not be material in amount or effect), in each case in accordance with generally accepted accounting principles consistently applied during the periods involved, except as may be noted therein. Neither Parent nor any of its Subsidiaries has any liabilities or obligations required to be disclosed in a balance sheet of Parent or in the notes thereto prepared in accordance with generally accepted accounting principles consistently applied except (a) liabilities or obligations reflected on, or reserved against in, a balance sheet of Parent or in the notes thereto, and included in the Parent Reports and (b) liabilities or obligations incurred since March 31, 1997 in the ordinary course of business.

- 4.7. Investigations; Litigation. Except as set forth in Section 4.7 of the Parent Disclosure Schedule, (a) to the knowledge of Parent, no material investigation by any governmental entity with respect to Parent or any of its Subsidiaries is pending (or, to Parent's knowledge, threatened) nor has any governmental entity indicated to Parent an intention to conduct the same; and (b) there are no actions, suits or proceedings pending against Parent or its Subsidiaries or, to the knowledge of Parent, threatened against Parent or its Subsidiaries, at law or in equity, or before or by any federal or state commission, board, bureau, agency or instrumentality, that are reasonably likely to have a Parent Material Adverse Effect.
- 4.8. Absence of Certain Changes. Since December 31, 1996, Parent has conducted its business only in the ordinary course of such business, and there has not been (i) any Parent Material Adverse Effect or any event which is reasonably likely to result in a Parent Material Adverse Effect; (ii) any declaration, setting aside or payment of any dividend or other distribution with respect to its capital stock (other than the regular quarterly cash dividends of \$0.40 per share, payable on Parent Common Stock); or (iii) any material change in its accounting principles, practices or methods.
- 4.9. Taxes. Parent and each of its Subsidiaries (i) have timely filed all material federal, state and foreign tax returns required to be filed by any of them for tax years ended prior to the date of this Agreement or requests for extensions have been timely filed and any such request shall have been granted and not expired, and all such returns are complete in all material respects, (ii) have paid or accrued all taxes shown to be due and payable on such returns and (iii) have properly accrued all such taxes for such periods subsequent to the periods covered by such returns.
- 4.10. Contracts. Except as disclosed in Section 4.10 of the Parent Disclosure Schedule, to the knowledge of Parent, with respect to Government Contracts, there is, as of the date hereof, no (i) civil fraud or criminal investigation by any government investigative agency that is reasonably likely to have a Parent Material Adverse Effect, (ii) suspension or debarment proceeding (or equivalent proceeding) against Parent or any of its Subsidiaries that is reasonably likely to have a Parent Material Adverse Effect, (iii) request by the government for a contract price adjustment based on a claimed disallowance by Defense Contract Audit Agency or claim of defective pricing in excess of \$120 million, (iv) dispute between Parent or any of its Subsidiaries and the U.S. Government which, since June 30, 1996, has resulted in a government contracting officer's determination and finding final decision where the amount in controversy exceeds or is expected to exceed \$120 million or (v) claim or equitable adjustment by Parent or any of its Subsidiaries against the U.S. Government or any third party in excess of \$120 million.

4.11. Employee Benefit Plans.

For the purpose of this Agreement, "Parent Benefit Plans" means all employee benefit plans and other benefit arrangements covering employees or former employees of Parent and its Subsidiaries and all employee agreements providing compensation, severance or other benefits to any employee or former employee of Parent or any of its Subsidiaries. With respect to each Parent Benefit Plan that is intended to be a "qualified plan" within the meaning of Section 401(a) of the Code, either the IRS has issued a favorable determination letter that has not been revoked, or an application for a favorable determination letter was timely submitted to the IRS for which no final action has been taken by the IRS. To the knowledge of Parent, there is no reason that is not susceptible to cure why the qualified status under Section 401(a) of the Code of any Parent Benefit Plan would be denied or revoked, whether retroactively or prospectively. Except as would not have a Parent Material Adverse Effect, no Parent Benefit Plan, any fiduciary thereof, nor Parent has incurred any liability or penalty under Section 4975 of the Code or Section 502(i) of ERISA. Except as would not have a Parent Material Adverse Effect, each Parent Benefit Plan has been maintained and administered in all material respects in compliance with its terms and with the ERISA, and the Code to the extent applicable thereto. Except as would not have a Parent Material Adverse Effect, neither Parent nor any ERISA Affiliate (during the period of its affiliated status and prior thereto, to its knowledge) has any existing liability currently due and payable that has not been satisfied in full under Title IV of ERISA or Section 412 of the Code. To the knowledge of Parent, there are no current plans to terminate, whether voluntarily or involuntarily, any materially underfunded pension plans of Parent or any ERISA Affiliate that are subject to Title IV of ERISA. Except as would not have a Parent Material Adverse Effect, to the knowledge of Parent, there are no pending or anticipated claims against or otherwise involving any of the Parent Benefit Plans and no suit, action or other litigation (excluding claims for benefits incurred in the ordinary course of the Parent Benefit Plan activities) has been brought against or with respect to any such Parent Benefit Plan, except for any of the foregoing which would not have a Parent Material Adverse Effect. All material contributions required to be made as of the date hereof to the Parent Benefit Plans have been made or provided for. Except as set forth in Section 4.11 of the Parent Disclosure Schedule, the execution of, and performance of the transactions contemplated in, this Agreement will not (either alone or upon the occurrence of any additional or subsequent events) constitute an event under any benefit plan, policy, arrangement or agreement or any trust or loan that will or may result in any payment (whether of severance pay or otherwise), acceleration, forgiveness of indebtedness, vesting, distribution, increase in benefits or obligation to fund benefits with respect to any employee.

4.12. No Brokers. Parent has not entered into any contract, arrangement or understanding with any person or firm which may result in the obligation of the Company or Parent to pay any finder's fee, brokerage or agent's commissions or other like payments in connection with the negotiations leading to this Agreement or the consummation of the transactions contemplated hereby, except that Parent has retained Bear, Stearns & Co. Inc. and Lehman Brothers as its financial advisors. Other than the foregoing arrangements, the Company is not aware of any claim for payment of any finder's fees, brokerage or agent's commissions or other like payments in connection with the negotiations leading to this Agreement or the consummation of the transactions contemplated hereby.

- 4.13. Opinion of Financial Advisor. Parent has received the opinion of Bear, Stearns & Co. Inc. to the effect that as of the date hereof, the consideration to be paid by Parent pursuant to the Merger is fair to Parent from a financial point of view.
- 4.14. Company Stock Ownership. Neither Parent nor any of its Subsidiaries owns any shares of Company Common Stock or other securities convertible into shares of Company Common Stock.
- 4.15. Pooling of Interests; Tax Reorganization. In the judgment of Parent, Parent and its Subsidiaries have not taken (or as of the date hereof failed to take) any action which would prevent the accounting for the Merger as a pooling of interests in accordance with APB No. 16, the interpretative releases issued pursuant thereto, and the pronouncements of the SEC. To the knowledge of the Parent, Parent has not taken or failed to take any action which would prevent the Merger from constituting a reorganization within the meaning of section 368(a) of the Code.
- 4.16. Environmental Matters. Except as described in Parent Reports, Parent and each of its Subsidiaries are in material compliance with all applicable Environmental Laws, except for instances of noncompliance that individually or in the aggregate do not, and, insofar as reasonably can be foreseen, in the future would not, have a Parent Material Adverse Effect. Such compliance includes, but is not limited to, the possession by Parent and its Subsidiaries of all material permits and other governmental authorizations required under applicable Environmental Laws, and compliance with the terms and conditions thereof. Neither Parent nor any of its Subsidiaries has received written notice of, or to the knowledge of Parent, is the subject of, any Environmental Claims that individually or in the aggregate would have a Parent Material Adverse Effect. To the knowledge of Parent, there are no circumstances that are reasonably likely to prevent or interfere with such material compliance in the future.

ARTICLE 5

COVENANTS

5.1. Alternative Proposals. The Company, its affiliates and their respective officers, directors, employees, representatives and agents shall immediately cease any existing discussions or negotiations, if any, with any parties conducted heretofore with respect to any acquisition of all or any material portion of the assets of, or any equity interest in, the Company or any of its Significant Subsidiaries or any business combination with the Company or any of its Significant Subsidiaries. The Company may, directly or indirectly, furnish information and access, in each case only in response to unsolicited requests therefor, to any corporation, partnership, person or other entity or group pursuant to confidentiality agreements, and may participate in discussions and negotiate with such entity or group concerning any merger, sale of material assets, sale of shares of capital stock or similar transaction involving the Company or any Significant Subsidiary (a "Transaction"), if such entity or group has submitted a written proposal to the Board relating to any such transaction (an "Alternative Proposal") and the Board by a majority vote determines in its good faith judgment, based as to legal matters on the advice of legal counsel, that failing to take such action would constitute a breach of the Board's fiduciary duty. The Board shall provide a copy of any such written proposal to Parent or Merger Sub immediately after receipt

thereof. Except as set forth above, neither the Company or any of its affiliates, nor any of its or their respective officers, directors, employees, representatives or agents, shall, directly or indirectly, encourage, solicit, participate in or initiate discussions or negotiations with, or provide any information to, any corporation, partnership, person or other entity or group (other than Parent and Merger Sub, any affiliate or associate of Parent and Merger Sub or any designees of Parent and Merger Sub) concerning any merger, sale of material assets, sale of shares of capital stock or similar transaction involving the Company or any Significant Subsidiary; provided, however, that nothing herein shall prevent the Board from taking, and disclosing to the Company's stockholders, a position contemplated by Rules 14d-9 and 14e-2 promulgated under the Exchange Act with regard to any tender offers, provided, further, that the Board shall not recommend that the stockholders of the Company tender their Shares in connection with any such tender offer unless the Board by a majority vote determines in its good faith judgment, based as to legal matters on the advice of legal counsel, that failing to take such action would constitute a breach of the Board's fiduciary duty. Nothing in this Section 5.1shall (x) permit the Company to terminate this Agreement (except as specifically provided in Article 7 hereof), (y) permit the Company to enter into any agreement with respect to a Transaction during the term of this Agreement (it being agreed that during the term of this Agreement, the Company shall not enter into any agreement with any person that provides for, or in any way facilitates, a Transaction (other than a confidentiality agreement in customary form)), or (z) affect any other obligation of the Company under this Agreement. Nothing in this Section 5.1 shall preclude the Company from consummating the transactions contemplated by the Logicon Merger Agreement.

5.2. Interim Operations of the Company.

- (a) Prior to the Effective Time, except as set forth in Section 5.2 of the Company Disclosure Schedule or as contemplated by any other provision of this Agreement and except as contemplated by the Logicon Merger Agreement, unless Parent has consented in writing thereto, the Company:
- (i) shall, and shall cause each of its Subsidiaries to, conduct its operations according to their usual, regular and ordinary course in substantially the same manner as heretofore conducted;
- (ii) shall use its reasonable efforts, and shall cause each of its Subsidiaries to use its reasonable efforts, to preserve intact their business organizations and goodwill, keep available the services of their respective officers and employees and maintain satisfactory relationships with those persons having business relationships with them;
- (iii) shall not, and shall cause its Subsidiaries not to, amend their respective Certificates of Incorporation or Bylaws or comparable governing instruments (other than amendments which are not material to the Company or to the consummation of the transactions contemplated by this Agreement);
- (iv) shall promptly notify Parent of (x) any material change in its condition (financial or otherwise), business, properties, assets, liabilities or the normal course of its business or of its properties, (y) any material litigation or material governmental complaints,

investigations or hearings (or communications indicating that the same may be contemplated), or (z) the breach of any representation or warranty contained herein:

- $% \left(v\right) =0$ (v) shall promptly deliver to Parent true and correct copies of any report, statement or schedule filed with the SEC subsequent to the date of this Agreement;
- (vi) shall not, and shall not permit any of its Subsidiaries to, except (x) in the ordinary course of business consistent with past practice, (y) as otherwise provided in this Agreement, enter into or amend any employment, severance or similar agreements or arrangements with any of their respective directors or executive officers;
- (vii) subject to the provisions of Section 5.1, shall not, and shall not permit any of its Subsidiaries to, authorize, propose or announce an intention to authorize or propose, or enter into an agreement with respect to, any merger, consolidation or business combination (other than the Merger), any acquisition of assets or securities, any disposition of assets or securities or any release or relinquishment of any contract rights in which, in any such case, the aggregate consideration for such transaction is in excess of \$100 million or which would have an adverse impact on the Company's ability to consummate the Merger;
- (viii) shall not, and shall not permit any of its Subsidiaries to, issue any shares of their capital stock or securities, except upon exercise of options outstanding as of the date hereof to purchase up to 914,000 shares of Company Common Stock under Company Stock Option Plans, or effect any stock split or otherwise change its capitalization;
- (ix) shall not, and shall not permit any of its Subsidiaries to, grant, confer or award any options, appreciation rights, warrants, conversion rights, restricted stock, stock units, performance shares or other rights, not existing on the date hereof, with respect to any shares of its capital stock or other securities of the Company or its Subsidiaries;
- (x) shall not, and shall not permit any of its Subsidiaries to, take any actions which would, or would be reasonably likely to, prevent the Merger from qualifying as a reorganization within the meaning of Section 368 of the Code;
- (xi) shall not, and shall not permit any of its Subsidiaries to, amend in any material respect the terms of the Company Benefit Plans, including, without limitation, any employment, severance or similar agreements or arrangements in existence on the date hereof, or adopt any new employee benefit plans, programs or arrangements or any employment, severance or similar agreements or arrangements;
- (xii) shall not, and shall not permit any of its Subsidiaries to, (x) incur, create, assume or otherwise become liable for borrowed money or assume, guarantee, endorse or otherwise become responsible or liable for the obligations of any other individual,

corporation or other entity or (y) make any loans or advances to any other person, except in each case in the ordinary course of business;

(xiii) shall not, and shall not permit any of its Subsidiaries to, make any material tax election other than in the ordinary course, or without the consent of Parent, which shall not unreasonably be withheld, settle or compromise any material tax liability;

(xiv) shall not, and shall not permit any of its Subsidiaries to, (y) declare, set aside or pay any dividend or make any other distribution or payment with respect to any shares of its capital stock or other ownership interests (other than regular quarterly cash dividends not to exceed \$0.40 per share of Company Common Stock and dividends and distributions from Subsidiaries of the Company to the Company or another of its Subsidiaries) or (z) directly or indirectly redeem, purchase or otherwise acquire any shares of its capital stock or capital stock of any of its Subsidiaries, or make any commitment for any such action; and

(xv) shall not, and shall not permit any of its Subsidiaries to, agree, in writing or otherwise, to take any of the foregoing actions or take any action which would make any representation or warranty in Article 3 hereof untrue or incorrect as of the Closing Date.

5.3. Interim Operations of Parent.

- (a) Prior to the Effective Time, except for the transactions currently being considered by Parent, which have been described to the Company (and which Parent shall not consummate prior to the Effective Time unless and until it shall have received a fairness opinion from an independent investment banking firm of national reputation reasonably satisfactory to the Company), or as contemplated by any other provision of this Agreement, unless the Company has consented in writing thereto, Parent:
- (i) shall not, and shall cause its Subsidiaries not to, amend their respective Certificates of Incorporation or Bylaws or comparable governing instruments (other than amendments which are not material to Parent or to the consummation of the transactions contemplated by this Agreement);
- (ii) shall promptly deliver to the Company true and correct copies of any report, statement or schedule filed with the SEC subsequent to the date of this Agreement;
- (iii) shall not, and shall not permit any of its Subsidiaries to, authorize, propose or announce an intention to authorize or propose, or enter into an agreement with respect to, any merger, consolidation or business combination (other than the Merger), any acquisition of assets or securities, any disposition of assets or securities or any release or relinquishment of any contract rights in which, in any such case, the aggregate consideration for such transaction is in excess of \$500 million or which would have an adverse impact on Parent's ability to consummate the Merger;
- (iv) shall promptly notify the Company of (x) any material change in its condition (financial or otherwise), business, properties, assets, liabilities or the normal course of its business or of its properties, (y) any material litigation or material

governmental complaints, investigations or hearings (or communications indicating that the same may be contemplated), or (z) the breach of any representation or warranty contained herein;

- (v) shall not, and shall not permit any of its Subsidiaries to, take any actions which would, or would be reasonably likely to, prevent the Merger from qualifying as a reorganization within the meaning of Section 368 of the Code;
- (vi) shall not declare, set aside or pay any dividend or make any other distribution or payment with respect to any shares of its capital stock (other than regular quarterly cash dividends not to exceed \$0.40 per share of Parent Common Stock and regular dividends on Parent Preferred Stock); and
- (vii) shall not, and shall not permit any of its Subsidiaries to, agree, in writing or otherwise, to take any of the foregoing actions or take any action which would make any representation or warranty in Article 4 hereof untrue or incorrect as of the Closing Date.
- 5.4. Meetings of Stockholders. Each of Parent and the Company will take all action necessary in accordance with applicable law and its Certificate of Incorporation and Bylaws to convene a meeting of its stockholders as promptly as practicable to consider and vote upon (i) in the case of Parent, the approval of the issuance of the shares of Parent Common Stock pursuant to the Merger contemplated hereby and (ii) in the case of the Company, the approval of this Agreement and the transactions contemplated hereby. The Board of Directors of each of Parent and the Company shall recommend such approval and Parent and the Company shall each take all lawful action to solicit such approval, including, without limitation, timely mailing the Proxy Statement/Prospectus (as defined in Section 5.9); provided, however, that such recommendation or solicitation shall not be required if and to the extent that the Board of Directors of Parent or the Company, as the case may be, determines, after the date hereof, and upon the advice of outside counsel, that the making of such recommendation or solicitation would involve a breach of its fiduciary duties to its stockholders imposed by law.
- 5.5. Filings; Other Actions. Subject to the terms and conditions herein provided, the Company and Parent shall: (a) promptly make their respective filings and thereafter make any other required submissions under the HSR Act with respect to the Merger; (b) use all reasonable efforts to cooperate with one another in (i) determining which other filings are required to be made prior to the Effective Time with, and which other consents, approvals, permits or authorizations are required to be obtained prior to the Effective Time from, governmental or regulatory authorities of the United States, the several states and foreign jurisdictions in connection with the execution and delivery of this Agreement and the consummation of the transactions contemplated hereby and (ii) timely making all such filings and timely seeking all such consents, approvals, permits or authorizations; and (c) use all reasonable efforts to take, or cause to be taken, all other action and do, or cause to be done, all other things necessary, proper or appropriate to consummate and make effective the transactions contemplated by this Agreement.
- 5.6. HSR Act. Subject to the following sentence, the parties shall take all actions necessary or appropriate to cause the prompt expiration or termination of any applicable waiting period under the HSR Act in respect of the Merger, including, without limitation, complying as

promptly as practicable with any requests for additional information. Without limiting the generality of the foregoing, if it is necessary in order to terminate the waiting period under the HSR Act or otherwise to permit the Closing to take place, Parent agrees to use all reasonable efforts to divest any assets held as of the date of this Agreement by Parent or the Company (or their Subsidiaries), to use all reasonable efforts to hold such assets separate pending such divestiture, or to enter into a consent decree requiring it to use all reasonable efforts to divest such assets, and to take such further action in connection therewith as may be necessary to enable the Closing to take place on or prior to March 31, 1998; provided that Parent shall not be required to take any action pursuant to this Section 5.6 if the taking of such action would have a significant adverse effect on the business, results of operations or financial condition of Parent and the Company (and their Subsidiaries), taken as a whole following the Effective Time.

- 5.7. Inspection of Records. From the date hereof to the Effective Time, each of the Company and Parent shall, and shall cause its Subsidiaries to, (i) allow all designated officers, attorneys, accountants and other representatives of the other party reasonable access at all reasonable times to its respective offices, records and files, correspondence, audits and properties, as well as to all information relating to its respective commitments, contracts, titles and financial position, or otherwise pertaining to its respective business and affairs, (ii) furnish to the other party and the other party's counsel, financial advisors, auditors and other authorized representatives such financial and operating data and other information as such persons may reasonably request and (iii) instruct its respective employees, counsel and financial advisors to cooperate with the other party in the other party's investigation of its respective business.
- 5.8. Publicity. The initial press release relating to this Agreement shall be a joint press release and thereafter the Company and Parent shall, subject to their respective legal obligations (including requirements of stock exchanges and other similar regulatory bodies), consult with each other, and use reasonable efforts to agree upon the text of any press release, before issuing any such press release or otherwise making public statements with respect to the transactions contemplated hereby and in making any filings with any federal or state governmental or regulatory agency or with any national securities exchange with respect thereto.
- 5.9. Registration Statement. Parent and the Company shall cooperate and promptly prepare and Parent shall file with the SEC as soon as practicable a Registration Statement on Form S-4 (the "Form S-4") under the Securities Act, with respect to the Parent Common Stock issuable in the Merger, which Registration Statement shall contain the joint proxy statement with respect to the meetings of the stockholders of the Company and of Parent in connection with the Merger (the "Proxy Statement/Prospectus"). The parties will cause the Proxy Statement/Prospectus, and Parent will cause the Form S-4, to comply as to form in all material respects with the applicable provisions of the Securities Act, the Exchange Act and the rules and regulations thereunder. Parent shall use all reasonable efforts, and the Company will cooperate with Parent, to have the Form S-4 declared effective by the SEC as promptly as practicable. Parent shall use its best efforts to obtain, prior to the effective date of the Form S-4, all necessary state securities law or "Blue Sky" permits or approvals required to carry out the transactions contemplated by this Agreement. The information supplied by the Company for inclusion or incorporation by reference in the Proxy Statement/Prospectus and the Form S-4 shall not (i) at the time the Form S-4 is declared effective, (ii) at the time the Proxy Statement/Prospectus (or any amendment thereof or supplement thereto) is first mailed to holders of Company Common Stock or

holders of Parent Common Stock, (iii) at the time of the Company Stockholders' Meeting or the Parent Stockholder's Meeting and (iv) at the Effective Time, contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein, in the light of the circumstances under which they are made, not misleading. The information supplied by Parent for inclusion or incorporation by reference in the Proxy Statement/Prospectus and the Form S-4 shall not (i) at the time the Form S-4 is declared effective, (ii) at the time the Proxy Statement/Prospectus (or any amendment thereof or supplement thereto) is first mailed to holders of Company Common Stock or holders of Parent Common Stock, (iii) at the time of the Company Stockholders' Meeting or the Parent Stockholder's Meeting and (iv) at the Effective Time, contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein, in the light of the circumstances under which they are made, not misleading. No amendment or supplement to the Proxy Statement/Prospectus will be made by the Company without the approval of Parent. Parent will advise the Company, promptly after it receives notice thereof, of the time when the Form S-4 has become effective or any supplement or amendment has been filed, the issuance of any stop order, the suspension of the qualification of the Parent Common Stock issuable in connection with the Merger for offering or sale in any jurisdiction, or any request by the SEC for amendment of the Proxy Statement/Prospectus or the Form S-4 or comments thereon and responses thereto or requests by the SEC for additional information.

- 5.10. Listing Application. Parent shall promptly prepare and submit to the NYSE a listing application covering the shares of Parent Common Stock issuable in the Merger, and shall use its best efforts to obtain, prior to the Effective Time, approval for the listing of such Parent Common Stock, subject to official notice of issuance.
- 5.11. Affiliate Letters. At least 30 days prior to the Closing Date, the Company shall deliver to Parent a list of names and addresses of those persons who were, in the Company's reasonable judgment, at the record date for its stockholders' meeting to approve the Merger, "affiliates" (each such person, an "Affiliate") of the Company within the meaning of Rule 145 of the rules and regulations promulgated under the Securities Act. The Company shall provide Parent such information and documents as Parent shall reasonably request for purposes of reviewing such list. The Company shall use all reasonable efforts to deliver or cause to be delivered to Parent, prior to the Closing Date, from each of the Affiliates of the Company identified in the foregoing list, an Affiliate Letter in the form attached hereto as Exhibit A. Parent shall be entitled to place legends as specified in such Affiliate Letters on the certificates evidencing any Parent Common Stock to be received by such Affiliates pursuant to the terms of this Agreement, and to issue appropriate stop transfer instructions to the transfer agent for the Parent Common Stock, consistent with the terms of such Affiliate Letters.
- 5.12. Expenses. Whether or not the Merger is consummated, all costs and expenses incurred in connection with this Agreement and the transactions contemplated hereby shall be paid by the party incurring such expenses except as expressly provided herein and except that the filing fee in connection with the filing of the Form S-4 or Proxy Statement/Prospectus with the SEC and the expenses incurred in connection with printing and mailing the Form S-4 and the Proxy Statement/Prospectus shall be shared equally by the Company and Parent.

5.13. Indemnity; Insurance. Parent agrees (i) that all rights to indemnification for acts or omissions occurring prior to the Effective Time in favor of the current or former directors or officers of the Company as provided in the Certificate of Incorporation or Bylaws of the Company shall survive the Merger and shall continue in full force and effect in accordance with their terms from the Effective Time of the Merger until the expiration of the applicable statute of limitations with respect to any claims against the current or former directors or officers of the Company arising out of such acts or omissions and (ii) to cause the Surviving Corporation to indemnify such current and former directors and officers in accordance with such rights of indemnification. For a period of six years after the Effective Time, Parent shall cause the Surviving Corporation to maintain officers' and directors' liability insurance for all persons currently covered under the Company's officers' and directors' liability insurance policies, in their capacities as officers and directors, on terms substantially no less advantageous to the covered persons than such existing insurance; provided, however, that Parent shall not be required in order to maintain or procure such coverage to pay an annual premium in excess of 150% of the current annual premium paid by the Company for its existing coverage (the "Cap"); and provided, further, that if equivalent coverage cannot be obtained, or can be obtained only by paying an annual premium in excess of 150% of the Cap, Parent shall only be required to obtain as much coverage as can be obtained by paying an annual premium equal to the Cap.

5.14. Employee Benefits.

- (a) For a period of two years following the Effective Time (or, in the case of employees covered by collective bargaining agreements, for the period required therein), Parent shall provide to persons who are employees of the Company or any of its Subsidiaries at the Effective Time (the "Company Personnel") employee benefit plans, programs and arrangements which in the aggregate are substantially comparable to those employee benefit plans, programs and arrangements generally provided to the employees of the Company or its Subsidiaries immediately prior to the Effective Time; provided, however, that subject to the foregoing, Parent shall not be precluded from amending or terminating any particular plan, program or arrangement.
- (b) Following the Effective Time, Parent shall cause the Company Benefit Plans to continue to recognize the service credit of the Company Personnel accrued as of the Effective Time under the Company Benefit Plans for purposes of participation, eligibility, vesting of benefits and benefit accrual, subject to the terms of such Company Benefit Plans, and such service credit shall also be recognized for purposes of participation, eligibility and vesting under any successor plans.
- (c) In the event of any change in coverage that applies generally to the Company Personnel during the two-year period following the Effective Time under any Company Benefit Plan that provides medical or health benefits, Parent shall cause such plan to recognize credit toward satisfying deductible expense requirements, out-of-pocket expense limits and maximum lifetime benefit limits of such Company Personnel or their eligible dependents, and to waive any pre-existing condition, exclusion or limitation, as and to the extent any such matter would previously have been recognized or waived (as the case may be) under the applicable Company Benefit Plan.

- (d) Parent shall cause the Surviving Corporation to honor in accordance with their terms and to perform the obligations of the Company under the severance and retention plans and agreements listed on Section 5.14 of the Company Disclosure Schedule.
- (e) Prior to the Effective time, the Company may (a) enter into retention agreements with corporate headquarters key employees to be identified and selected by the Company after consultation with Parent; provided that the maximum amount payable under all such retention agreements in the aggregate shall not exceed \$15 million; and (b) adopt a general retention program for corporate headquarter employees; provided that the maximum amount payable under such program in the aggregate shall not exceed \$18 million; and provided further, however, that if the employees are covered by more than one Company retention or severance agreement, plan or program, an offset shall be applied to avoid any duplication of benefits.
- (f) Prior to the Effective Time, the Company shall cause to be amended each of the Company Benefit Plans (and/or related funding arrangements) listed on Section 5.14 of the Company Disclosure Schedule to eliminate all provisions that would become effective upon a "Change in Control" of the Company (or any similar triggering event), as and to the extent such amendments are consistent with requirements of law.
- (g) Following the Effective Time, Parent shall cause Company Personnel who have previously been granted Company stock options to be eligible to participate in the Parent stock option plan. At the first meeting of the Compensation Committee of the Parent Board of Directors following the Effective Time at which stock options are granted to Parent employees generally, Parent shall cause a recommendation to be made to the Compensation Committee to grant options under the Parent stock option plan to such Company Personnel on a basis that takes into consideration the Company stock options previously granted to such Company Personnel. If the Effective Time occurs after the first such meeting of the Compensation Committee, in 1998 or thereafter, Parent shall present its recommendation for granting Parent stock options in accordance with the foregoing at the next regularly scheduled meeting of the Compensation Committee or sooner if schedules and agendas permit.
- 5.15 Reorganization. From and after the date hereof and until the Effective Time, neither Parent nor the Company nor any of their respective subsidiaries or other affiliates shall (i) knowingly take any action, or knowingly fail to take any action, that would jeopardize qualification of the Merger as a reorganization within the meaning of Section 368(a) of the Code; or (ii) enter into any contract, agreement, commitment or arrangement with respect to the foregoing.
- 5.16. Shareholder Rights Plan. The Company shall take all action necessary to render the Rights Agreement, dated September 21, 1988 between Northrop Corporation and Manufacturers Hanover Trust Company (the "Shareholder Rights Plan") and the rights issued pursuant to the Shareholder Rights Plan inapplicable to the transactions contemplated hereby.

5.17. [Reserved]

5.18. Agreements. Between the date hereof and the Closing Date, neither Parent nor the Company shall enter into any agreement which Parent or the Company, as the case may be, knows or has reason to know is reasonably likely to cause any major customer of Parent or the

Company (or their respective subsidiaries) to terminate any material contracts, agreements or other obligations that exist between that customer on the one hand, and Parent, the Company (or Parent and the Company following the Merger) or any subsidiary of either, on the other hand and Parent and the Company shall take all reasonable action appropriate to an effort to avoid such termination.

- 5.19. Parent Board of Directors. The Company shall be entitled to designate for appointment or election to Parent's Board of Directors following consummation of the Merger, three persons to be mutually agreed upon by Parent and the Company. Parent's Board of Directors shall take all necessary action to cause Kent Kresa to be elected as Vice Chairman of Parent's Board of Directors following the Merger.
- 5.20. Takeover Statute. If any "fair price," "moratorium," "control share acquisition" or other form of antitakeover statute or regulation shall become applicable to the transactions contemplated hereby, the Company and the members of the Board of Directors of the Company shall grant such approvals and take such actions as are reasonably necessary so that the transactions contemplated hereby may be consummated as promptly as practicable on the terms contemplated hereby and otherwise act to eliminate or minimize the effects of such statute or regulation on the transactions contemplated hereby.

ARTICLE 6

CONDITIONS

- 6.1. Conditions to Each Party's Obligation to Effect the Merger. The respective obligation of each party to effect the Merger shall be subject to the fulfillment at or prior to the Closing Date of the following conditions:
- (a) This Agreement and the transactions contemplated hereby shall have been approved by the holders of the issued and outstanding shares of capital stock of the Company and the Parent Stockholder Approval shall have been obtained.
- (b) The waiting period applicable to the consummation of the Merger under the HSR $\,$ Act shall have expired or been terminated.
- (c) Neither of the parties hereto shall be subject to any order or injunction of a court of competent jurisdiction in the United States which prohibits the consummation of the transactions contemplated by this Agreement. In the event any such order or injunction shall have been issued, each party agrees to use its best efforts to have any such injunction lifted.
- (d) The Form S-4 shall have become effective and shall be effective at the Effective Time, and no stop order suspending effectiveness of the Form S-4 shall have been issued, no action, suit, proceeding or investigation by the SEC to suspend the effectiveness thereof shall have been initiated and be continuing, and all material approvals under state securities laws relating to the issuance or trading of the Parent Common Stock to be issued to the Company stockholders in connection with the Merger shall have been received.

- (e) The Parent Common Stock to be issued to the Company stockholders in connection with the Merger shall have been approved for listing on the NYSE, subject only to official notice of issuance.
- (f) All consents, authorizations, orders and approvals of (or filings or registrations with) any governmental commission, board or other regulatory body required in connection with the execution, delivery and performance of this Agreement shall have been obtained or made, except for filings in connection with the Merger and any other documents required to be filed after the Effective Time and except where the failure to have obtained or made any such consent, authorization, order, approval, filing or registration would not have a material adverse effect on the business of Parent and the Company (and their respective Subsidiaries), taken as a whole, following the Effective Time.
- 6.2. Conditions to Obligation of the Company to Effect the Merger. The obligation of the Company to effect the Merger shall be subject to the fulfillment at or prior to the Closing Date of the following conditions:
- (a) Parent shall have performed in all material respects its agreements contained in this Agreement required to be performed on or prior to the Closing Date, the representations and warranties of Parent and Merger Sub contained in this Agreement and in any document delivered in connection herewith shall be true and correct in all material respects as of the Closing Date, except that those representations and warranties which address matters only as of a particular date shall have been true and correct as of such date, and the Company shall have received a certificate of the President or a Vice President of Parent, dated the Closing Date, certifying to such effect.
- (b) The Company shall have received the opinion of Fried, Frank, Harris, Shriver & Jacobson, special counsel to the Company, to the effect that the Merger will be treated for federal income tax purposes as a reorganization within the meaning of Section 368(a) of the Code.
- (c) The Company shall have received a letter of its independent public accountants, dated the Effective Time, in form and substance reasonably satisfactory to it, stating that such accountants concur with management's conclusion that the Merger will qualify as a transaction to be accounted for in accordance with the pooling of interests method of accounting under the requirements of APB No. 16, provided that if such accountants are not able to deliver such letter due to a transaction of the type currently being considered by Parent, which has been described to the Company, such letter shall not be a condition to the Company's obligations hereunder.
- (d) From the date of this Agreement through the Effective Time, there shall not have occurred a Parent Material Adverse Effect.
- 6.3. Conditions to Obligation of Parent and Merger Sub to Effect the Merger. The obligations of Parent and Merger Sub to effect the Merger shall be subject to the fulfillment at or prior to the Closing Date of the following conditions:

- (a) The Company shall have performed in all material respects its agreements contained in this Agreement required to be performed on or prior to the Closing Date, the representations and warranties of the Company contained in this Agreement and in any document delivered in connection herewith shall be true and correct in all material respects as of the Closing Date, except that those representations and warranties which address matters only as of a particular date shall have been true and correct as of such date, and Parent shall have received a certificate of the President or a Vice President of the Company, dated the Closing Date, certifying to such effect.
- (b) Parent shall have received the opinion of King & Spalding, special counsel to Parent, dated the Closing Date, to the effect that the Merger will be treated for Federal income tax purposes as a reorganization within the meaning of Section 368(a) of the Code, and that the Company and Parent will each be a party to that reorganization within the meaning of Section 368(b) of the Code.
- (c) Parent shall have received a letter of its independent public accountants, dated the Effective Time, in form and substance reasonably satisfactory to it, stating that such accountants concur with management's conclusion that the Merger will qualify as a transaction to be accounted for in accordance with the pooling of interests method of accounting under the requirements of APB No. 16, provided that if such accountants are not able to deliver such letter due to a transaction of the type currently being considered by Parent which has been described to the Company, such letter shall not be a condition to Parent's obligations hereunder.
- (d) From the date of this Agreement through the Effective Time, there shall not have occurred a Company Material Adverse Effect.

ARTICLE 7

TERMINATION

- 7.1. Termination by Mutual Consent. This Agreement may be terminated and the Merger may be abandoned at any time prior to the Effective Time, before or after the approval of this Agreement by the stockholders of the Company, by the mutual consent of Parent and the Company.
- 7.2. Termination by Either Parent or the Company. This Agreement may be terminated and the Merger may be abandoned by action of the Board of Directors of either Parent or the Company if (a) the Merger shall not have been consummated by March 31, 1998, or (b) the approval of the Company's stockholders required by Section 6.1(a) shall not have been obtained at a meeting duly convened therefor or at any adjournment thereof, or (c) the approval of Parent's stockholders required by Section 6.1(a) shall not have been obtained at a meeting duly convened therefor or at any adjournment thereof, or (d) a United States federal or state court of competent jurisdiction or United States federal or state governmental, regulatory or administrative agency or commission shall have issued an order, decree or ruling or taken any other action permanently restraining, enjoining or otherwise prohibiting the transactions contemplated by this Agreement and such order, decree, ruling or other action shall have become final and nonappealable; provided, that

the party seeking to terminate this Agreement pursuant to this clause (d) shall have used all reasonable efforts to remove such injunction, order or decree; and provided, in the case of a termination pursuant to clause (a) above, that the terminating party shall not have breached in any material respect its obligations under this Agreement in any manner that shall have proximately contributed to the failure to consummate the Merger by March 31, 1998.

- 7.3. Termination by the Company. This Agreement may be terminated and the Merger may be abandoned at any time prior to the Effective Time, before or after the adoption and approval by the stockholders of the Company referred to in Section 6.1(a), by action of the Board of Directors of the Company, if (a) in the exercise of its good faith judgment as to fiduciary duties to its stockholders imposed by law based on the written opinion of outside counsel, the Board of Directors of the Company determines that such termination is required by reason of an Alternative Proposal being made, or (b) there has been a breach by Parent or Merger Sub of any representation or warranty contained in this Agreement that has had or will have a Parent Material Adverse Effect, which breach is not curable or, if curable, is not cured within 30 days after written notice of such breach is given by the Company to Parent, or (c) there has been a material breach of any of the covenants or agreements set forth in this Agreement on the part of Parent, which breach is not curable or, if curable, is not cured within 30 days after written notice of such breach is given by the Company to Parent. Notwithstanding the foregoing, the Company's ability to terminate this Agreement pursuant to Section 7.2 or this Section 7.3 is conditioned upon the prior payment by the Company of any amounts owed by it pursuant to Section 7.5(a).
- 7.4. Termination by Parent. This Agreement may be terminated and the Merger may be abandoned at any time prior to the Effective Time, before or after the approval by the stockholders of Parent referred to in Section 6.1(a), by action of the Board of Directors of Parent, if (a) the Board of Directors of the Company shall have (i) withdrawn or modified in a manner materially adverse to Parent its approval or recommendation of this Agreement or the Merger or (ii) recommended an Alternative Proposal to the Company stockholders, or (b) there has been a breach by the Company of any representation or warranty contained in this Agreement that has had or will have a Company Material Adverse Effect, which breach is not curable or, if curable, is not cured within 30 days after written notice of such breach is given by the Company to Parent, or (c) there has been a material breach of any of the covenants or agreements set forth in this Agreement on the part of the Company, which breach is not curable or, if curable, is not cured within 30 days after written notice of such breach is given by Parent to the Company.

7.5. Effect of Termination and Abandonment.

(a) If this Agreement is terminated by the Company or Parent pursuant to Section 7.2(b) or 7.4(a) (i), and, prior to such termination, (x) a proposal with respect to a Transaction shall have been made, and (y) within six (6) months after such termination, the Company enters into any agreement with respect to a Transaction, or any third party shall acquire beneficial ownership of 50.1% or more of the Company's outstanding shares of voting stock, then within two business days after the execution of such an agreement or the consummation of such acquisition (whichever shall first occur), the Company shall pay Parent, by wire transfer of immediately available funds, a fee (the "Termination Fee") of \$200 million. If this Agreement is terminated by the Company pursuant to clause (a) of Section 7.3 or by Parent pursuant to clause (a) (ii) of Section 7.4, then the Company shall pay Parent the Termination Fee, which fee shall be

payable by wire transfer of same day funds either at the time contemplated in the last sentence of Section 7.3 if applicable or, otherwise, within two business days after such termination. The Company acknowledges that the agreements contained in this Section 7.5(a) are an integral part of the transactions contemplated in this Agreement, and that, without these agreements, Parent and Merger Sub would not enter into this Agreement; accordingly, if the Company fails to promptly pay the amount due pursuant to this Section 7.5(a), and, in order to obtain such payment, Parent or Merger Sub commences a suit which results in a judgment against the Company for the fee set forth in this Section 7.5(a), the Company shall pay to Parent its costs and expenses (including attorneys' fees) in connection with such suit, together with interest on the amount of the fee at the rate of 12% per annum from the date such fee was required to be paid.

- (b) In the event of termination of this Agreement and the abandonment of the Merger pursuant to this Article 7, all obligations of the parties hereto shall terminate, except the obligations of the parties set forth in this Section 7.5 and Section 5.12 and except for the provisions of Sections 8.3, 8.4, 8.6, 8.8, 8.9, 8.12, 8.13 and 8.14 and except for the Confidentiality Agreement previously executed between the Company and Parent (the "Confidentiality Agreement"). Moreover, in the event of termination of this Agreement pursuant to Section 7.2, 7.3 or 7.4, nothing herein shall prejudice the ability of the nonbreaching party from seeking damages from any other party for any wilful breach of this Agreement, including without limitation, attorneys' fees and the right to pursue any remedy at law or in equity.
- 7.6. Extension; Waiver. At any time prior to the Effective Time, any party hereto, by action taken by its Board of Directors, may, to the extent legally allowed, (a) extend the time for the performance of any of the obligations or other acts of the other parties hereto, (b) waive any inaccuracies in the representations and warranties made to such party contained herein or in any document delivered pursuant hereto and (c) waive compliance with any of the agreements or conditions for the benefit of such party contained herein. Any agreement on the part of a party hereto to any such extension or waiver shall be valid only if set forth in an instrument in writing signed on behalf of such party.

ARTICLE 8

GENERAL PROVISIONS

- 8.1. Nonsurvival of Representations, Warranties and Covenants. All representations, warranties and agreementscovenants in this Agreement or in any instrument delivered pursuant to this Agreement shall not survive the Merger, provided, however, that the agreementscovenants contained in Article 2, the last sentence of Section 5.135.11, Section 5.12, Section 5.13, Section 5.14, Section 5.19 and this Article 8 shall survive the Merger, but not beyond the extent, if any, specified therein. Nothing in this Section 8.1 shall affect Section 7.5(b) of this Agreement.
- 8.2. Notices. Any notice required to be given hereunder shall be sufficient if in writing, and sent by facsimile transmission and by courier service (with proof of service), hand delivery or certified or registered mail (return receipt requested and first-class postage prepaid), addressed as follows:

If to Parent or Merger Sub:

Lockheed Martin Corporation 6801 Rockledge Drive Bethesda, Maryland 20817 Attention: Frank H. Menaker, Jr., Esq. Senior Vice President and General Counsel

with copies to:

Dewey Ballantine 1301 Avenue of the Americas New York, New York 10019 Attention: William J. Phillips, Esq.

If to the Company:

Northrop Grumman Corporation 1840 Century Park East Los Angeles, California 90067 Attention: Richard R. Molleur, Esq. Corporate Vice President and General Counsel

with copies to:

Fried, Frank, Harris, Shriver & Jacobson One New York Plaza
New York, New York 10004-1980
Attention: Charles M. Nathan, Esq.

or to such other address as any party shall specify by written notice so given, and such notice shall be deemed to have been delivered as of the date so telecommunicated, personally delivered or mailed.

- 8.3. Assignment; Binding Effect. Neither this Agreement nor any of the rights, interests or obligations hereunder shall be assigned by any of the parties hereto (whether by operation of law or otherwise) without the prior written consent of the other parties. Subject to the preceding sentence, this Agreement shall be binding upon and shall inure to the benefit of the parties hereto and their respective successors and assigns. Notwithstanding anything contained in this Agreement to the contrary, except for the provisions of Section 5.13, nothing in this Agreement, expressed or implied, is intended to confer on any person other than the parties hereto or their respective heirs, successors, executors, administrators and assigns any rights, remedies, obligations or liabilities under or by reason of this Agreement.
- 8.4. Entire Agreement. This Agreement, the Exhibits, the Company Disclosure Schedule, the Parent Disclosure Schedule, the Confidentiality Agreement and any documents delivered by the parties in connection herewith constitute the entire agreement among the parties with respect to the subject matter hereof and supersede all prior agreements and understandings among the parties with respect thereto. No addition to or modification of any provision of this

Agreement shall be binding upon any party hereto unless made in writing and signed by all parties hereto.

- 8.5. Amendment. This Agreement may be amended by the parties hereto, by action taken by their respective Boards of Directors, at any time before or after approval of matters presented in connection with the Merger by the stockholders of the Company and Parent, but after any such stockholder approval, no amendment shall be made which by law requires the further approval of stockholders without obtaining such further approval. This Agreement may not be amended except by an instrument in writing signed on behalf of each of the parties hereto.
- 8.6. Governing Law. This Agreement shall be governed by and construed in accordance with the laws of the State of Delaware without regard to its rules of conflict of laws. Each of the Company and Parent hereby irrevocably and unconditionally consents to submit to the exclusive jurisdiction of the courts of the State of Delaware and of the United States of America located in the State of Delaware (the "Delaware Courts") for any litigation arising out of or relating to this Agreement and the transactions contemplated hereby (and agrees not to commence any litigation relating thereto except in such courts), waives any objection to the laying of venue of any such litigation in the Delaware Courts and agrees not to plead or claim in any Delaware Court that such litigation brought therein has been brought in an inconvenient forum.
- 8.7. Counterparts. This Agreement may be executed by the parties hereto in separate counterparts, each of which when so executed and delivered shall be an original, but all such counterparts shall together constitute one and the same instrument. Each counterpart may consist of a number of copies hereof each signed by less than all, but together signed by all of the parties hereto.
- 8.8. Headings. Headings of the Articles and Sections of this Agreement are for the convenience of the parties only, and shall be given no substantive or interpretive effect whatsoever.
- 8.9. Interpretation. In this Agreement, unless the context otherwise requires, words describing the singular number shall include the plural and vice versa, and words denoting any gender shall include all genders and words denoting natural persons shall include corporations and partnerships and vice versa.
- 8.10. Waivers. Except as provided in this Agreement, no action taken pursuant to this Agreement, including, without limitation, any investigation by or on behalf of any party, shall be deemed to constitute a waiver by the party taking such action of compliance with any representations, warranties, covenants or agreements contained in this Agreement. The waiver by any party hereto of a breach of any provision hereunder shall not operate or be construed as a waiver of any prior or subsequent breach of the same or any other provision hereunder.
- 8.11. Incorporation of Exhibits. The Company Disclosure Schedule, the Parent Disclosure Schedule and all Exhibits attached hereto and referred to herein are hereby incorporated herein and made a part hereof for all purposes as if fully set forth herein.

- 8.12. Severability. Any term or provision of this Agreement which is invalid or unenforceable in any jurisdiction shall, as to that jurisdiction, be ineffective to the extent of such invalidity or unenforceability without rendering invalid or unenforceable the remaining terms and provisions of this Agreement or affecting the validity or enforceability of any of the terms or provisions of this Agreement in any other jurisdiction. If any provision of this Agreement is so broad as to be unenforceable, the provision shall be interpreted to be only so broad as is enforceable.
- 8.13. Enforcement of Agreement. The parties hereto agree that irreparable damage would occur in the event that any of the provisions of this Agreement was not performed in accordance with its specific terms or was otherwise breached. It is accordingly agreed that the parties shall be entitled to an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions hereof in any Delaware Court, this being in addition to any other remedy to which they are entitled at law or in equity.
- 8.14. Subsidiaries. As used in this Agreement, the word "Subsidiary" when used with respect to any party means any corporation or other organization, whether incorporated or unincorporated, of which such party directly or indirectly owns or controls at least a majority of the securities or other interests having by their terms ordinary voting power to elect a majority of the board of directors or others performing similar functions with respect to such corporation or other organization, or any organization of which such party is a general partner. When a reference is made in this Agreement to Significant Subsidiaries, the words "Significant Subsidiaries" shall refer to Subsidiaries (as defined above) which constitute "significant subsidiaries" under Rule 405 promulgated by the SEC under the Securities Act.

IN WITNESS WHEREOF, the parties have executed this Agreement and caused the same to be duly delivered on their behalf on the day and year first written above.

LOCKHEED MARTIN CORPORATION

Ву	:	
		Bennett Vice President Financial Officer

HURRICANE SUB, INC.

By:

Marcus C. Bennett
President

NORTHROP GRUMMAN CORPORATION

By:
Richard B. Waugh, Jr.

Richard B. Waugh, Jr. Corporate Vice President and Chief Financial Officer

FORM OF AFFILIATE LETTER

Lockheed Martin Corporation 6801 Rockledge Drive Bethesda, Maryland 20817

Ladies and Gentlemen:

I have been advised that as of the date of this letter I may be deemed to be an "affiliate" of Northrop Grumman Corporation, a Delaware corporation (the "Company"), as the term "affiliate" is (i) defined for purposes of paragraphs (c) and (d) of Rule 145 of the rules and regulations (the "Rules and Regulations") of the Securities and Exchange Commission (the "Commission") under the Securities Act of 1933, as amended (the "Act"), or (ii) used in and for purposes of Accounting Series, Releases 130 and 135, as amended, of the Commission. Pursuant to the terms of the Agreement and Plan of Merger dated as of July 2, 1997 (the "Agreement"), between Lockheed Martin Corporation, a Delaware corporation ("Parent"), Hurricane Sub, Inc., a Delaware corporation and a wholly owned subsidiary of Parent ("Merger Sub"), and the Company, Merger Sub will be merged with and into the Company (the "Merger").

As a result of the Merger, I will receive shares of Common Stock, par value \$1.00 per share, of Parent (the "Parent Securities") in exchange for shares owned by me of Common Stock, par value \$1.00 per share, of the Company.

I represent, warrant and covenant to Parent that in the event I receive any Parent Securities as a result of the Merger:

- A. I shall not make any sale, transfer or other disposition of the Parent Securities in violation of the Act or the Rules and Regulations.
- B. I have carefully read this letter and the Agreement and discussed the requirements of such documents and other applicable limitations upon my ability to sell, transfer or otherwise dispose of the Parent Securities to the extent I felt necessary, with my counsel or counsel for the Company.
- C. I have been advised that the issuance of Parent Securities to me pursuant to the Merger has been registered with the Commission under the Act on a Registration Statement on Form S-4. However, I have also been advised that, since at the time the Merger was submitted for a vote of the stockholders of the Company, I may be deemed to have been an affiliate of the Company and the distribution by me of the Parent Securities has not been registered under the Act, I may not sell, transfer or otherwise dispose of the Parent Securities issued to me in the Merger unless (i) such sale, transfer or other disposition has been registered under the Act, (ii) such sale,

transfer or other disposition is made in conformity with Rule 145 promulgated by the Commission under the Act, or (iii) in the opinion of counsel reasonably acceptable to Parent, or pursuant to a "no action" letter obtained by the undersigned from the staff of the Commission, such sale, transfer or other disposition is otherwise exempt from registration under the Act.

- D. I understand that Parent is under no obligation to register the sale, transfer or other disposition of the Parent Securities by me or on my behalf under the Act or to take any other action necessary in order to make compliance with an exemption from such registration available.
- E. I also understand that stop transfer instructions will be given to Parent's transfer agents with respect to the Parent Securities and that there will be placed on the certificates for the Parent Securities issued to me, or any substitutions therefor, a legend stating in substance:

"THE SHARES REPRESENTED BY THIS CERTIFICATE WERE ISSUED IN A TRANSACTION TO WHICH RULE 145 PROMULGATED UNDER THE SECURITIES ACT OF 1933 APPLIES. THE SHARES REPRESENTED BY THIS CERTIFICATE MAY ONLY BE TRANSFERRED IN ACCORDANCE WITH THE TERMS OF AN AGREEMENT DATED _____ BETWEEN THE REGISTERED HOLDER HEREOF AND LOCKHEED MARTIN CORPORATION, A COPY OF WHICH AGREEMENT IS ON FILE AT THE PRINCIPAL OFFICES OF LOCKHEED MARTIN CORPORATION."

F. I also understand that unless the transfer by me of my Parent Securities has been registered under the Act or is a sale made in conformity with the provisions of Rule 145, Parent reserves the right to put the following legend on the certificates issued to my transferee:

"THE SHARES REPRESENTED BY THIS CERTIFICATE HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933 AND WERE ACQUIRED FROM A PERSON WHO RECEIVED SUCH SHARES IN A TRANSACTION TO WHICH RULE 145 PROMULGATED UNDER THE SECURITIES ACT OF 1933 APPLIES. THE SHARES HAVE BEEN ACQUIRED BY THE HOLDER NOT WITH A VIEW TO, OR FOR RESALE IN CONNECTION WITH, ANY DISTRIBUTION THEREOF WITHIN THE MEANING OF THE SECURITIES ACT OF 1933 AND MAY NOT BE SOLD, PLEDGED OR OTHERWISE TRANSFERRED EXCEPT IN ACCORDANCE WITH AN EXEMPTION FROM THE REGISTRATION REQUIREMENTS OF THE SECURITIES ACT OF 1933."

It is understood and agreed that the legends set forth in paragraphs E and F above shall be removed by delivery of substitute certificates without such legend if such legend is not required for purposes of the Act or this Agreement. It is understood and agreed that such legends and the stop orders referred to above will be removed if (i) one year shall have elapsed from the date the undersigned acquired the Parent Securities received in the Merger and the provisions of

Rule 145(d)(2) are then available to the undersigned, (ii) two years shall have elapsed from the date the undersigned acquired the Parent Securities received in the Merger and the provisions of Rule 145(d)(3) are then available to the undersigned, or (iii) Parent has received either an opinion of counsel, which opinion and counsel shall be reasonably satisfactory to Parent, or a "no action" letter obtained by the undersigned from the staff of the Commission, to the effect that the restrictions imposed by Rule 145 under the Act no longer apply to the undersigned.

I further represent to and covenant with Parent that I will not sell, transfer or otherwise dispose of any Parent Securities received by me in the Merger or any other shares of the capital stock of Parent until after such time as results covering at least 30 days of combined operations of the Company and Parent have been published by Parent, in the form of a quarterly earnings report, an effective registration statement filed with the Commission, a report to the Commission on Form 10-K, 10-Q or 8-K, or any other public filing or announcement which includes such combined results of operations. Parent shall notify the "affiliates" of the publication of such results. Notwithstanding the foregoing, I understand that I will not be prohibited from selling up to 10% of the Parent Securities received by me in the Merger during the aforementioned period.

Execution of this letter should not be considered an admission on my part that I am an "affiliate" of the Company as described in the first paragraph of this letter or as a waiver of any rights I may have to object to any claim that I am such an affiliate on or after the date of this letter.

	Very truly yours,
	Name:
Accepted this day of , 1997 by LOCKHEED MARTIN CORPORATION	
Ву:	
 Name: Title:	